Tax planning and financial default: role of corporate life cycle

Alessandro Gabrielli and Giulio Greco Department of Economics and Management, University of Pisa, Pisa, Italy

Abstract

Purpose – Drawing on the resource-based view (RBV), this study investigates how tax planning affects the likelihood of financial default in different stages of the corporate life cycle.

Design/methodology/approach – Collecting a large sample of US firms between 1989 and 2016, hypotheses are tested using a hazard model. Several robustness and endogeneity checks corroborate the main findings. **Findings** – The results show that tax-planning firms are less likely to default in the introduction and decline stages, while they are more likely to default in the growth and maturity stages. The findings suggest that introductory and declining firms use cash resources obtained from tax planning efficiently to meet their needs and acquire other useful resources. In growing and mature firms, tax aggressiveness generates unnecessary slack resources, weakens managerial discipline and increases reputational risks.

Practical implications – The results shed light on the benefits and costs associated with tax planning throughout firms' life cycle, holding great significance for managers, investors, lenders and other stakeholders. **Originality/value** – This study contributes to the literature that examines resource management at different life cycle stages by showing that cash resources from tax planning are managed in distinctive ways in each life cycle stage, having a varied impact on the likelihood of default. The authors shed light on underexplored cash resources. Furthermore, this study shows the potential linkages between the agency theory and RBV.

Keywords Tax planning, Corporate life cycle, Financial default, Resource management Paper type Research paper

1. Introduction

Tax planning [1] is a key managerial activity with relevant implications for firms' growth, internationalization, corporate value and capital structure (Cooper and Nguyen, 2020; Wang *et al.*, 2020). The resource-based view (RBV) posits that tax planning is an important cash-generating activity (Hasan *et al.*, 2017; Medioli *et al.*, 2020; Magerakis, 2022). Studies suggest that, although not inimitable, cash is one of the most desirable resources for value creation because it is flexible, readily convertible and can be used to acquire strategic resources (Lee and Wu, 2016; Warnier *et al.*, 2013). Its scarcity hinders firms from not only acquiring other resources but also managing existing resources, increasing the risk of default (Trahms *et al.*, 2013). Other researchers assert that tax planning is a value-destroying activity because it generates slack cash resources, diminishing the managerial discipline over efficient investing

© Alessandro Gabrielli and Giulio Greco. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) licence. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this licence may be seen at http://creativecommons.org/licences/by/4.0/legalcode

This research was partially carried out when Alessandro Gabrielli was a Visiting Ph.D. Student at the WHU– Otto Beishem School of Management in Vallendar, Germany. The authors are grateful to the members of the WHU– Otto Beishem School of Management Brown Bag and Tax Readings Group and to the participants of the SIDREA–EAA Workshop jointly held by the Italian Society of Accounting and Business Economics Educators and the European Accounting Association, as well as to the participants of EAA 2022 Annual Congress and the Ninth International Conference of JIAR. The authors are also thankful to Martin Jacob, Lisa Hillmann, Petra Andries, and Marco Maria Mattei for providing helpful comments. Alessandro Gabrielli gratefully acknowledges financial support from the Department of Economics and Management at the University of Pisa.

Tax planning and financial default

321

Received 23 August 2022 Revised 8 March 2023 7 June 2023 4 September 2023 Accepted 18 September 2023



Management Decision Vol. 61 No. 13, 2023 pp. 321-355 Emerald Publishing Limited 0025-1747 DOI 10.1108/MD-07-2022-0928 and usage of resources and increasing reputational risks (Daniel *et al.*, 2004; Lee and Wu, 2016; Nohria and Gulati, 1996).

RBV scholars suggest exploring how a firm obtains and manages resources throughout its life cycle (Sirmon *et al.*, 2011; Trahms *et al.*, 2013; Zahra, 2021). Tax planning is likely heterogeneous across a firm's life cycle. Notably, the life cycle affects firms' *ex ante* (pretax planning) resources, their propensity to engage in tax planning (Hasan *et al.*, 2017) and how they allocate and use resources generated from tax planning (Cooper and Nguyen, 2020; Dickinson, 2011). Hence, whether tax planning increases or decreases the likelihood of financial default likely depends on the tax-planning firm's life cycle stage. Therefore, this study investigates how tax planning influences the likelihood of financial default when moderated by the firm's life cycle stage.

This research is important for two reasons. First, it answers recent calls for research on resource management at different life cycle stages (Zahra, 2021), as well as on firm characteristics and outcomes of tax planning (Wang *et al.*, 2020). Second, academics and practitioners are interested in understanding how tax planning affects the likelihood of financial default in those life cycle stages where access to resources is constrained, such as the introduction and decline stages (Zahra, 2021).

Using a large sample of US public firms in the period 1989–2016, the research regresses tax planning, measured by effective tax rates (ETRs) and ETRs volatility, on a financial default-dependent variable. Following prior studies (Altman, 1968; Beaver *et al.*, 2010), we define financial default as the firms filing under either Chapter 11 or Chapter 7 procedures. Our moderator variable is the corporate life cycle. Following prior management literature, we operationalize life cycle using the Dickinson cash flow patterns model (Cuypers *et al.*, 2016; Sánchez *et al.*, 2017; Hasan and Cheung, 2018; Hsu, 2018). The research adds a set of control variables—identified in prior literature on financial default—and year and industry fixed effects. Our study also performs a battery of tests to address potential endogeneity concerns related to reverse causality and omitted variable bias (Wooldridge, 2010).

The findings show that the association between tax planning and financial default varies at different life cycle stages. We find that aggressive tax planner firms (i.e. firms with lower ETRs) are less likely to default in the introductory and decline life cycle stages. The results are economically significant. Using hazard ratios, we find that a 1% reduction of ETRs lowers—*ceteris paribus*—the probability of a default of 6.65% and 3.51% for firms in the decline and introductory stages of the corporate life cycle, respectively. These findings suggest that firms in the introductory and decline stages efficiently use cash resources from tax planning to meet their financial needs and acquire resources useful for their survival. Aggressive tax planning, however, leads to a higher default likelihood among firms in the growth and mature stages—a 1% increase in the ETRs volatility increases the default likelihood by 1.92% in growth firms and by 2.36% in mature firms. Volatile tax rates, which are visible to external stakeholders through financial reports, may signal generation of unnecessary slack resources, lack of managerial discipline and increased reputational risks (Duong *et al.*, 2022).

This study contributes to literature in several ways. First, as stated earlier, it answers the call for research on resource management at specific life cycle stages (Zahra, 2021). Our findings show that cash resources from tax planning are managed and allocated in distinctive ways across life cycle stages, thereby having a varied impact on the default likelihood. Moreover, while most prior studies focus on valuable, rare, inimitable and organizationally embedded (VRIO) resources, we contribute to the literature on less explored cash resources (Warnier *et al.*, 2013; Magerakis, 2022).

Second, this study contributes to prior RBV literature with insights on potential linkages between agency theory and RBV, at a conceptual and empirical level (Biscotti *et al.*, 2018; Zona *et al.*, 2018; Purkayastha *et al.*, 2021; Luu, 2023). This study shows that cash slack from

MD

61.13

aggressive tax planning increases agency costs and weakens the optimal management of the firm's resources at certain life cycle stages. In this sense, RBV and agency theory may be complementary in explaining why tax planning increases the default risk in growth and mature firms, thus offering a more refined explanation of the phenomenon.

Third, this study extends the literature on tax planning. It shows that firms' life cycle stages explain not only the heterogeneity of approaches to tax planning (Hasan *et al.*, 2017; Wang *et al.*, 2020; Cooper and Nguyen, 2020) but also the outcome of such activities. Compared to Hasan *et al.*'s (2017) findings, our findings indicate that a firm's life cycle stage must be considered to not only understand why certain firms are more tax aggressive than others, but also gain insight into why only certain firms seem to benefit from tax planning.

Fourth, this study contributes to the literature on how life cycle-specific management decisions affect the likelihood of default. Previous studies have shown that a firm's life cycle stage moderates the relationship between the likelihood of default and several management decisions—such as investment, R&D and CSR (Al-Hadi *et al.*, 2019; Anandarajan *et al.*, 2010; Koh *et al.*, 2015). This study extends our current understanding of this aspect by exploring the moderating role of firms' life cycle stages in the relationship between default likelihood and tax planning. In this sense, it has the potential to complement the recent literature on firm's failure and survival under specific conditions (Islam and Fatema, 2023; Paeleman *et al.*, 2023) and in specific life cycle stages (Mota *et al.*, 2022; Ehsani and Osiyevskyy, 2023).

The findings present several managerial implications. The research informs managers and stakeholders on the costs and benefits of tax planning across a firm's life cycle stages as well as the stakeholders' assessment of tax planning. At certain life cycle stages, tax planning acts as a signal to outsiders for increased managerial discretion and reputational risks (Duong *et al.*, 2022), thus affecting the default likelihood. In this respect, we provide a quantitative estimate of the magnitude of the tax planning effect on financial default likelihood, which is of interest to managers and stakeholders.

The remainder of the paper is structured as follows. Section 2 reviews the relevant literature and develops hypotheses. Section 3 describes the research methodology. Section 4 presents the results. Section 5 critically discusses the research results, highlighting their implications to theory and management practice. Section 6 concludes the paper.

2. Literature review and hypotheses development

2.1 Literature review

The RBV suggests that firms are "a collection of productive resources, where the choice of different uses of these resources over time is determined by administrative decision" (Penrose, 1959, p. 21). Numerous RBV-based empirical studies have examined how firms' resource management [2] impacts their survival and performance (Andersén, 2011; Barney, 1991; Newbert, 2007; Warnier *et al.*, 2013; Wernerfelt, 1984) [3]. Drawing on the RBV, researchers have integrated the concept of the corporate life cycle into the analysis of firms' resources (Sirmon *et al.*, 2011). The management literature identifies five distinct phases in the corporate life cycle: introduction, growth, maturity, shake-out and decline (Miller and Friesen, 1984). These stages are characterized by differences in the business environment, resource availability, strategy and decision-making (Miller and Friesen, 1984).

Recent studies indicate that different life cycle stages are associated with different resource management and commitment choices and significant changes in resource endowment (Sirmon *et al.*, 2011; Trahms *et al.*, 2013; Zahra, 2021) [4]. Some studies have demonstrated different resource management in different life cycle stages, focusing on default avoidance. For example, Koh *et al.* (2015) showed that firms in the introduction stage change their CEOs more quickly and reduce dividend payouts to avoid bankruptcy. With lower or less appealing investment opportunities, mature firms engage in corporate social

Tax planning and financial default

responsibility to attract resources and reduce their risk of default (Al-Hadi et al., 2019). Declining firms attempt to avoid or delay financial defaults by boosting internal efficiency 61.13 and gaining support from key stakeholders (Anandarajan et al., 2010; Morrow et al., 2007; Trahms et al., 2013).

> This study focuses on a notable managerial decision, tax planning and its effect on the likelihood of default. Efficient tax planning reduces financial "scarce resources" that are transferred outside the firm and increases firm performance (Watson, 2015). In managerial decision-making, tax planning is likely a wider evaluation of a firm's economic fundamentals. cash flow patterns and internal resource assessment (Scholes et al., 2014; Xu and Zheng, 2020). Hasan et al. (2017) show that firms' propensity to engage in tax planning varies based on their life cycle stage; introductory and declining firms are, on average, more active than growing and mature firms. However, whether tax planning increases or reduces the likelihood of default based on the life cycle stage remains unexplored.

> Prima facie, tax planning is a salient cash-generating activity that limits the transfer of financial resources from the firm to the government and generates high discretion slack [5] (George, 2005). The life cycle stage affects two key conditions underlying firms' tax planning decision: the *ex ante* availability of resources (and the related) financial needs, and the managers' commitment or discretion regarding using corporate resources (Dickinson, 2011; Sirmon *et al.*, 2011).

2.2 Hypotheses development

In the introduction stage, firms generally have negative operating cash flows. Some do not even break-even and need resources for investments (Dickinson, 2011; Maiti et al., 2020; Song et al., 2022; Franczak et al., 2023). Since they are run by owner-entrepreneurs, these firms usually have centralized organizational structures (Miller and Friesen, 1984; Franczak et al., 2023) and limited separation between ownership and control (Badertscher et al., 2013; Fama and Jensen, 1983). Resource constraints and limited access to external financing sources increase their propensity to obtain additional resources through tax planning (Edwards et al., 2016; Higgins et al., 2015; Cerrato et al., 2023). They seek and use legal tax minimization opportunities, including benign deferral tax strategies and tax loss deductions, that are unlikely to be challenged by tax authorities (Hasan et al., 2017).

By using resources shrewdly and having little or no managerial discretion (Miller and Friesen, 1984; Sirmon et al., 2011), cash resources generated through tax planning likely foster firms' competitive advantage and performance, reducing the risk of financial default. Owing to their small size and low exposure to capital markets, reputational costs associated with tax planning are limited for these firms (Al-Hadi et al., 2016; Zimmerman, 1983). Hence, in the introduction stage, tax planning likely provides firms with an additional source of cash resources and is negatively associated with their financial default. Thus, we formulate the following hypothesis.

H1. Tax planning is negatively associated with the financial default of firms in the introductory stage of their life cycle.

In the growth stage, firms have robust operating cash flows, raise cash from equity and/or leverage financing to fund investments. Tax planning is less appealing as a source of generating cash (Hasan et al., 2017). At this stage, there is a greater degree of separation between ownership and control with decentralization and functionalization (Filatotchev *et al.*, 2006; Miller and Friesen, 1984). Furthermore, slack resources generated through tax planning can diminish managerial discipline over efficient investment (Nohria and Gulati, 1996). Managers may exploit their discretion and use slack resources to continue projects having negative present value, make suboptimal investments, or take up excessively costly and

324

MD

risky R&D activities (George, 2005; Lee and Wu, 2016; Nohria and Gulati, 1996). Shareholders are likely to perceive tax planning activities as risky and a sign of managerial wealth diversion, penalizing firms with a higher cost of capital (Desai *et al.*, 2007; Duong *et al.*, 2022). Lenders also penalize high-tax-planning firms with a high cost of debt (Dhawan *et al.*, 2020). Based on these arguments, we posit the following hypothesis.

H2. Tax planning is positively associated with the financial default of firms in the growth stage of their life cycle.

Once firms exhaust all growth opportunities, they mature. Mature firms are older, larger, known in the market, and investors and lenders perceive them as less risky (Easley and O'Hara, 2004; Miller and Friesen, 1984). These firms have abundant operating cash flow owing to reduced investment needs and negative cash flow from financing activities due to debt repayment and cash dividend payouts (Dickinson, 2011; Faff et al., 2016), Resource slack and complex bureaucratic structures can increase managerial discretion in how funds obtained from tax planning are employed (Quinn and Cameron, 1983; Nohria and Gulati, 1996; Carnes et al., 2017). Considering high resource availability, poor investment opportunities and high managerial discretion, tax planning likely becomes ancillary to managerial wealth extraction and/or limited benefit to shareholders and bondholders. For mature firms, as Flanagan and O'Shaughnessy (2005, p. 445) noted, "reputation is perhaps one of its most important strategic resources," and they focus on reinforcing their relationship with the external environment (Boyd et al., 2010; Jacob et al., 2021; Zimmerman, 1983). Meanwhile, aggressive tax planning can raise doubts from the government and tax authorities (Gallemore *et al.*, 2014). Customers may also be unwilling to support aggressive tax-planning companies (Hardeck and Hertl, 2014). Moreover, investors and lenders do not expect tax avoidance from mature and risk-averse firms and can react adversely to such tax planning activities by increasing the cost of capital and debt (Hasan et al., 2017). Concerns regarding reputational costs and penalties may outweigh the expected economic benefits from tax planning. Thus, we posit the following hypothesis.

H3. Tax planning is positively associated with the financial default of firms in the maturity stage of their life cycle.

The corporate life cycle ends in the decline stage. When firms enter this stage, growth rates, operating cash flows and investment expenditures plummet (Hasan *et al.*, 2017). They divest and engage in asset restructuring activities (e.g. sale of assets) to pay off debt, renegotiate interest rates and continue operating (Dickinson, 2011; Koh *et al.*, 2015). The lack of resources and close monitoring by shareholders and external lenders substantially reduce managerial discretion over resource usage (Dickinson, 2011; Jensen, 1986). Then, tax planning can be a valuable source of cash and business restructuring to avoid bankruptcy. In declining firms, the benefits of survival overcome the risks of tax avoidance for both managers and shareholders (i.e. litigation, penalty and reputational risks) (Hasan *et al.*, 2017; Koh *et al.*, 2015; Morrow *et al.*, 2007; Richardson *et al.*, 2015). Based on these arguments, we formulate the following hypothesis.

H4. Tax planning is negatively associated with the financial default of firms in the decline stage of their life cycle.

3. Method

3.1 Sample

We collected data from the Compustat North America database. We downloaded financial statement data of both active and inactive US firms. Then, we merged it with bankruptcy

Tax planning and financial default

MD information gathered from the CRSP, which provides information about the reasons for a firm's delisting (delisting codes: 400 "Liquidation" and 574 "Bankruptcy"). Initially, our sample comprised 283,633 firm-year observations pertaining to the period between 1989 and 2016. After excluding observations with incomplete data, the final sample had 90,790 firm-year observations. Our sample period starts in 1989 because Compustat provides cash flow data of all firms starting from this year (Dickinson, 2011). The sample period ends in 2016 because former President Donald Trump enacted a major change in the US tax system, and the implementation of the Tax Cuts and Jobs Act began in 2016. Since variation in the statutory tax rate may affect both dependent and independent variables, we limited our analysis to investigating a period that was not substantially affected by tax rate variations at the federal level.

3.2 Research design

We used two estimation approaches to test our hypotheses: a hazard model and a logit model. First, we use a hazard model. Scholars suggest that it is an appropriate estimation strategy for analyzing firms' financial default because it can incorporate time-variant features of model covariates (Bandopadhyaya, 1994; Shumway, 2001). Since financial default is dynamic in nature, the hazard model can provide a robust estimation of the probability that a firm will default at a given point in time (Esteve-Pérez and Mañez-Castillejo, 2008; Shumway, 2001). Next, we use the logit model as a robustness test. The logit model is a standard approach in examining financial defaults (Altman *et al.*, 2019; Campbell *et al.*, 2008). Following Hasan *et al.* (2017), we estimate our model (Equation 1) using two approaches: (a) a pooled regression on the full sample and (b) a separate regression on the subsample of firms in each life cycle stage.

$$\begin{aligned} Default_{it} = & \beta_0 + \beta_1 (Tax \ Planning)_{i,t-1} + \beta_2 (Introduction)_{i,t} + \beta_3 (Tax \ Planning)_{i,t-1} x (Introduction)_{i,t} + \\ & \beta_2 (Growth)_{i,t} + \beta_3 (Tax \ Planning)_{i,t-1} x (Growth)_{i,t} + \beta_4 (Mature)_{i,t} + \\ & \beta_5 (Tax \ Planning)_{i,t-1} x (Mature)_{i,t} + \beta_6 (Decline)_{i,t} + \beta_6 (Tax \ Planning)_{i,t-1} x (Decline)_{i,t} + \\ & \beta_7 (WC_TA)_{i,t-1} + \beta_8 (RE_TA)_{i,t-1} + \beta_9 (EBIT_TA)_{i,t-1} + \beta_{10} (MV_TL)_{i,t-1} + \beta_{11} (S_TA)_{i,t-1} + \\ & \sum_{t=0}^n \text{Year } Fixed \ Effects - \text{Industry } Fixed \ Effects + \varepsilon_{i,t} \end{aligned}$$

Default_{it} is the measure of financial default. It is a binary variable that equals one for each firm-year observation of a firm filed under Chapters 11 or 7. Otherwise, it equals zero. Following previous research, we measured financial default as the firm's decision to start the legal procedure of Chapters 11 or 7 (Kücher *et al.*, 2020; Lussier, 1995) [6]. *Tax Planning*_{*i*,*t*-1} is the independent variable. Following previous research, we measure tax planning using ETRs. Specifically, we use *GAAP ETR* as the first proxy, calculated as income tax expense (*txt*) on pretax income (*pi*). In addition to *GAAP ETR*, we use a long-term five-year standard deviation of *ETRs* (*GAAP ETRVol*). The *GAAP ETRVol* captures the risk inherent in corporate tax positions (Drake *et al.*, 2019; Guenther *et al.*, 2017). Blouin (2014) argues that combining these two proxies can provide a true and complete picture of corporate tax planning and the level of its aggressiveness. We also ensure that tax volatility is related to tax planning and not driven by the firm's pretax performance volatility by controlling for the firm's performance in the base models and endogeneity checks.

The corporate life cycle is the moderator variable. $Lifecycle_{i,t}$ is a dummy variable that identifies the life cycle stage when the firm defaults. Following previous studies (Hasan *et al.*, 2017; Koh *et al.*, 2015), we proxied corporate life cycle stages using firms' cash flow patterns, similar to Dickinson's model (2011). Dickinson (2011) argues that a firm's progression from

one life cycle stage to another is not a function of its age but of its cash flow patterns. Cash flow patterns have a robust relationship with life cycle fundamentals and explain why not all firms progress in the same order despite their age (Dickinson, 2011; Josefy et al., 2017) We adopt the Dickinson model because it provides a less erroneous identification of the corporate life cycle with respect to other classification criteria that are free from distributional assumptions (Hsu, 2018). Although alternative identification strategies exist (e.g. Anthony and Ramesh, 1992), it is a common stand in the literature that they are less effective or more erroneous than the Dickinson model (Habib and Hasan, 2019). According to the recent literature on the corporate life cycle, firms' transition from one stage to another cannot be fully explained by the age or the dynamics of their retained earnings (Dickinson, 2011; Yang et al., 2021). Therefore, a classification based on cash flow patterns can provide a consolidated and theoretically consistent criterion, limiting the researcher's arbitrariness in their identification (i.e. distributional assumption bias) (Dickinson, 2011). Several studies in management, accounting and organization science have used (or built on) Dickinson's model to identify life cycle stages (e.g. Cuypers et al., 2016; Hasan and Cheung, 2018; Hsu, 2018; Sánchez et al., 2017). To ensure the robustness of our results, in all of our models, we also control for the level of retained earnings in total assets (RE TA), a well-established measure in the financial default literature (Altman et al., 2019) and used by DeAngelo et al. (2006) as a proxy for the life cycle stage.

Following Dickinson (2011), we define corporate life cycle stages as follows:

Introduction: if *OANCF* < 0, *INVCF* < 0 and *FINCF* > 0,

Growth: if OANCF > 0, IVNCF < 0 and FINCF > 0,

Maturity: if OANCF > 0, INVCF < 0 and FINCF < 0,

Decline: if *OANCF*<0, *INVCF*>0 and *FINCF* \leq or = > 0 and

Shake-out: the remaining firm-years (those that are not classified as introduction, growth, maturity, or decline) [7].

Given its residual nature, previous studies have not considered the shake-out stage in their analyses or included it only as a benchmark (Hasan *et al.*, 2017; Koh *et al.*, 2015). Accordingly, we did not consider this stage in our analyses.

As control variables, we include these financial ratios based on the literature on financial defaults (Altman et al., 2019; Altman, 1968): working capital/total assets ("WC_TA"), retained earnings/total assets ("RE TA"), earnings before interest and taxes/total assets ("EBIT TA"), market value of equity/total liabilities ("MV_TL") and sales/total assets ("S_TA"). WC TA denotes the level of net liquid assets in the total capital invested and is a measure of corporate liquidity. RE TA represents cumulative profitability implicitly related to a firm's age (DeAngelo et al., 2006). On average, the RE_TA ratio increases as the firm ages. EBIT_ TA represents firms' profitability. It specifically reflects the part of corporate profitability that arises from firms' productivity, net of any leverage or tax effect. It is important to control for it because it permits us to parse out the effect of tax avoidance on financial default while limiting the risk that negative corporate performance could affect the interpretation of our results. MV TL is a measure of market performance calculated as the sum of the market value of common and ordinary shares divided by the total liabilities. Finally, the S_TA (assetturnover ratio) measures the total assets that contribute to the firm's sales-generating activities (Altman, 1968). All the models include industry and year fixed effects and standard errors clustered by firms, which control for firm-level fixed effects (Petersen, 2009).

We use lagged independent and control variables. Using lagged dependent variables can help mitigate endogeneity concerns regarding reverse causality (Vergara, 2010), providing a

Tax planning and financial default

MD more conservative examination of the potential endogeneity bias. All the variables were winsorized at 1%.

4. Results

4.1 Descriptive statistics

Table 1 displays the descriptive statistics of all variables with *Default* as the dependent variable. There are 2,121 defaulted observations, 2.3% of the total firm-year observations. The non-defaulted observations amount to 88,669, accounting for 97.6% of the total firm-year observations.

Our explanatory variable is tax planning, proxied by *GAAP ETR* and *GAAP ETRVol*. The mean *GAAP ETR* is 0.332 (33.2%), 0.336 (33.6%) and 0.331 (33.1%) for the total, defaulted and non-defaulted samples, respectively. The mean *GAAP ETRVol* is 0.048 (4.8%), 0.053 (5.3%) and 0.047 (4.7%) for the total, defaulted and non-defaulted samples, respectively. The *t*-statistics show that the mean-in-difference of tax-planning proxies in the defaulted and non-defaulted samples is highly significant.

Descriptive statistics of control variables suggest that, on average, firms filing under Chapters 7 or 11 have a low *RE_TA*, *EBIT_TA* and *MV_TL* and a high *WC_TA* and *S_TA*.

For the total, defaulted and non-defaulted samples, the mean of WC_TA is 0.244 (24.4%), 0.273 (27.3%) and 0.240 (24.0%), respectively. The mean of RE_TA in the total, defaulted and non-defaulted samples is 0.146 (14.6%), 0.101 (10.1%) and 0.153 (15.3%), respectively. The mean of $EBIT_TA$ in the total, defaulted and non-defaulted samples is 0.100 (10.0%), 0.089 (8.9%) and 0.102 (10.2%), respectively. The mean of MV_TL is 3.477 (347.7%), 3.530 (353.0%) and 3.091 (309.1%) in the total, non-defaulted and defaulted samples, respectively. Finally, the mean of S_TA is 1.113 (111.3%), 1.096 (109.6%) and 1.235 (123.5%) in the whole, non-defaulted and defaulted samples, respectively. The mean-indifference of all control variables in the defaulted and non-defaulted samples is statistically significant.

4.2 Univariate analysis

Table 2 shows the Spearman's correlation coefficients. Univariate analysis shows that *GAAP ETR* and *GAAP ETRVol* are positively and highly associated with *Default*. Additionally, *Default* is highly correlated with all life cycle stages, except for the growth and shake-out stages, at the <0.01 significance level. More specifically, *Default* is highly and positively correlated with the introduction and decline stages but highly and negatively correlated with the maturity phase. Furthermore, all control variables are significantly associated with *Default* in the expected direction.

4.3 Main results

Table 3 reports the results of employing the hazard model. For the sake of brevity, Table 3 reports only the coefficients of tax planning and the interaction term of tax planning and life cycle. Across all models in Table 3, the control variables are significantly associated with *Default* in the expected direction (not shown).

Column 1 in Table 3 shows that *GAAP ETR* is significantly and negatively associated with *Default*. This finding suggests that tax planning results in more default since lower tax rates imply a greater likelihood of default. Column 2 shows that *GAAP ETRVol* is positively associated with *Default*. These findings support the notion that tax planning is associated with more default when firm's profitability, size, leverage, liquidity and retained earnings are controlled for. They are consistent with the findings of previous studies (e.g. Dhawan *et al.*, 2020; Noga and Schnader, 2013). Columns 3 and 4 add the dummies of life cycle stages (not

I		SD	.095	211 232 690 573	and	I	Tax planning
		575	.395 0 .068 0	404 0 397 0 147 0 531 3 547 0	ferences		and financia defaul
	otal 30,790)	p25 I	.283 0.	0.069 0 0.066 0 0.065 0. 1.837 4. .649 1.	an-in-dif		
	T_0 (N = 9	dian J	357 0 032 0	224 0 212 0 826 0 034 0 034 0	ludes me		329
		an Me	32 0.: 18 0.0	49055 11000 11000	e also inc		
		Mea	0.33 0.04	0.24 0.10 0.10 3.47 1.11	he table		
	hDiff	t- statistics	(-3.94) (-12.52)	(-5.10) (25.18) (57.09) (19.02) (-25.75)	89–2016. TI		
	Mean-ir	Diff	-0.00423	-0.0102^{****} 0.115^{****} 0.0472^{****} 0.637^{****} -0.138^{****}	g the period 198		
		SD	$0.102 \\ 0.052$	$\begin{array}{c} 0.214\\ 0.388\\ 0.388\\ 0.056\\ 3.499\\ 0.587\end{array}$	s during		
	s	p75	$0.404 \\ 0.079$	$\begin{array}{c} 0.441\\ 0.319\\ 0.32\\ 0.132\\ 3.710\\ 1.763\end{array}$: analyse		
	ilted firm = 2,121)	p25	$0.279 \\ 0.014$	$\begin{array}{c} 0.107\\ 0.036\\ 0.053\\ 0.053\\ 0.826\\ 0.760\end{array}$	led in ou		
	Defaı (n	Median	0.367 0.036	$\begin{array}{c} 0.265\\ 0.153\\ 0.087\\ 1.609\\ 1.193\end{array}$	bles includ		
		Mean	$0.336 \\ 0.053$	$\begin{array}{c} 0.273\\ 0.101\\ 0.089\\ 3.091\\ 1.235 \end{array}$	ain varia		
		SD	$0.094 \\ 0.046$	$\begin{array}{c} 0.210\\ 0.392\\ 0.049\\ 3.713\\ 0.569\end{array}$	all the m used		
	rms	p75	$0.393 \\ 0.066$	$\begin{array}{c} 0.398\\ 0.406\\ 0.149\\ 4.632\\ 1.514\end{array}$	istics for ariables		
	faulted fi = 88,669)	p25	$0.283 \\ 0.013$	0.065 0.072 0.067 0.837 0.636	tive stat control v		
	Non-de (n =	Median	$0.356 \\ 0.031$	$\begin{array}{c} 0.219\\ 0.222\\ 0.101\\ 1.863\\ 1.014\end{array}$	ts descrip dent and o uthors		
	stics	Mean	$0.331 \\ 0.047$	$\begin{array}{c} 0.240\\ 0.153\\ 0.102\\ 3.530\\ 1.096\end{array}$	table repoi e indepen ated by a		
	Summary statis		GAAP ETR _{i,t} GAAP	$E_{IKV0i,(t;t-4)}$ $WC_{TA_{it}}$ $RE_{TA_{it}}$ $EBIT_{TA_{it}}$ $MV_{TL_{it}}$ $S TA_{i}$,	Note(s): This t t-statistics of th Source(s): Cre		Table 1 Descriptive statistice

MD 61.13	c	$TA_{i,t}$				Ι	Mature, on total	
	SS	$MV_{-}TL_{\rm i,t}$				$\frac{1}{-0.0333^{^{*6+8}}}$	bilities, Sales	
330	rol variable	$TA_{i,t}$				${1 \atop 0.272^{^{ m Merst}}}{0.249^{^{ m Merst}}}$	t). Introduct on total lia	
	Cont	$TA_{i,t}$				$\begin{array}{c}1\\0.172^{***}\\0.0463^{****}\\0.00947^{*}\end{array}$	o <i>ETRVoli</i> _{i(it} alue of equity	
	Can	$TA_{\rm i,t}$				$egin{array}{c} I \\ 0.0799^{****} \\ 0.128^{****} \\ 0.531^{****} \\ 0.215^{*****} \end{array}$	ETR _{i.} , GAAI ets, Market v	
	voidance	$ETRVol_{i,(t,t-4)}$			I	0.0123**** -0.186*** -0.190*** -0.0195*** -0.0783***	<i>Default; GAAP</i> txes on total ass	
	Taxav	$ETR_{i,t}$			$I - 0.114^{****}$	$\begin{array}{c} -0.0440^{****} \\ 0.122^{****} \\ 0.0637^{****} \\ -0.0464^{****} \\ 0.173^{****} \end{array}$	lyses, namely: interest and ta	
		Decline		1	-0.0174^{***} 0.0474^{****}	$\begin{array}{c} 0.0584^{****} \\ -0.0701^{****} \\ -0.164^{****} \\ 0.00299 \\ -0.0112^{***} \end{array}$	ded in the ana arnings before rely	
		Shake-out		1 -0.00117	-0.000761 0.00817^{*}	$\begin{array}{c} 0.00182 \\ -0.0271^{***} \\ -0.00822^{*} \\ 0.00474 \\ -0.0157^{****} \end{array}$	variables inclu total assets, Ed level, respecti	
	Life cycle	Mature		$rac{1}{-0.0053}$	-0.00804^{*} -0.0641^{***}	$\begin{array}{c} -0.0910^{***}\\ 0.0895^{***}\\ 0.163^{***}\\ -0.0938^{***}\\ 0.108^{***}\end{array}$	een the main l earnings on , 5 and 10% .	
		Growth		$\begin{array}{c}1\\-0.705^{****}\\-0.00714\\-0.0868^{****}\end{array}$	0.0182^{***} 0.00940^{*}	$\begin{array}{c} -0.0290^{***}\\ -0.0189^{***}\\ -0.0378^{***}\\ 0.0815^{***}\\ -0.132^{****}\end{array}$	relations betw ssets, Retainec mce at the 1%	
		Introduction		$\begin{array}{c}1\\-0.174^{^{9966}}\\-0.233^{^{9966}}\\-0.0236\\-0.0236\end{array}$	0.0185^{***} 0.0380^{***}	0.133 ^{****} -0.0475 ^{****} -0.0819 ^{****} -0.00435 0.0974 ^{****}	e Spearman cor pital on total a istical signific	
	ation matrix	$Default_{i,t}$	I	0.142**** -0.00158 -0.0733**** -0.00367 0.0397****	0.0146^{****} 0.0456^{****}	0.0516**** -0.0433**** -0.0869**** -0.0391**** 0.0799****	uble reports the ne, Working ca represent stat teed by authors	
Table 2. Correlation coefficients	Spearman correl:	Covariates	$Default_{i,t}$	Life cycle Introduction Growth Mature Shake-out Decline	Tax avoidance GAAP ETR _{i,t} GAAP ETRVol _{i,(t1-4)}	Control variables WC_TA _{it} RE_TA _{it} EBIT_TA _{it} BIT_TA _{it} MV_TL _{it} S_TA _{it}	Note(s): This ta Shake-out, Declir assets. ***, ***, ***, Source(s): Crea	

(1) -0.6794a -0.6676a -0.6676a -0.6676a -0.6676a -0.6676a -0.6676a -0.6676a -0.651a -0.652a	$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$
fib (+) 2.86 ³⁶⁴ 2.34 ⁷⁶⁴ (530) (+) (0.31) (0.31) (0.30) (530) (+) (0.31) (0.31) (0.30) (530) (+) (0.31) (0.31) (0.30) (530) (+) (0.31) (0.31) (0.30) (500) (+) (0.31) (0.30) (0.30) (500) (+) (0.30) (0.30) (0.30) (500) (+) (-) (0.30) (0.30) (0.30) (+) (-) (0.30) (0.30) (0.30) (+) (-) (0.30) (0.30) (0.30) (+) (-) (0.30) (0.30) (0.30) (+) (-) (-) (0.30) (0.30) (+) (-) (-) (-) (-) (+) (-) (-) (-) (-) (+) (-) (-) (-) (-) (+) (-) <td< td=""><td>offet 236° and 53 <th< td=""></th<></td></td<>	offet 236° and 53 236° and 53 <th< td=""></th<>
(1) (1) (1) (2) (1) (1) (1) (1) (1) (2) (1) (1) (1) (1) (1) (2) (1) (1) (2) (1) (1) (2) (1) (1) (2) (1) (1) (2) (1) (1) (2) (2) (2) (2) (2) (2) (2) (2) (2) (1) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) (2) ($ \begin{array}{c c c c c c c c c c c c c c c c c c c $
(1) (1) (1) (1) (1) (1) (1) (1)	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $
(1) (1) (1) (1) (1) (1) (1) (1)	() () () () () () () () () () () () () (
(1) (1) (1) (1) (1) (1) (1) (1)	(+) (-) (-) (-) (-) (-) (-) (-) (-) (-) (-
(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)	$ \begin{array}{llllllllllllllllllllllllllllllllllll$
(+) (+) (+) (-) (-) (-) (-) (-) (-) (-) (-	(+) (-) (-) (-) (-) (-) (-) (-) (-) (-) (-
(+) (-) (-) (-) (-) (-) (-) (-) (-	(1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
(-) (-) 90,790	(-) $\begin{array}{c} 0.0031\\ 0.0790\\ WD\\ WD\\ WD\\ WD\\ WD\\ WD\\ WD\\ WD\\ WD\\ WD$
90,790 90,790 90,790 90,790 90,790 90,790 NO NO NO YES YES YES YES YES YES YES YES YES YES	$\begin{array}{c c c c c c c c c c c c c c c c c c c $
	estimated coefficients obtained by regressing the hazard model with a dependent binary variable ("Default") identifying firms entering Chapter 11 (restructuring) or Chapter 7 1.3 and 5 report the results obtained by including a 1-year lagged <i>GAAP ETR</i> variable, calculated by taking the total income tax expense at time t-1 on the pre-tax accounting Heitzman, 2010). The higher the <i>GAAP ETR</i> is, the lower the level of tax avoidance is. Columns 2, 4 and 6 report the results obtained by using a 1-year lagged <i>GAAP ETRVol</i> lard deviation of <i>GAAP ETR</i> from time t-1 to time t-5. The higher the <i>GAAP ETRVol</i> is, the higher the level of risk associated to the firm's tax position is. Columns 1 and 2 report at deviation of <i>GAAP ETR</i> from time t-1 to time t-5. The higher the <i>GAAP ETRVol</i> is, the higher the level of risk associated to the firm's tax position is. Columns 1 and 2 report at deviation of <i>GAAP ETR</i> from time t-1 to time t-5. The higher the <i>GAAP ETRVol</i> is, the higher the level of risk associated to the firm's tax position is. Columns 1 and 2 report at deviation of <i>GAAP ETR</i> from time t-1 to time t-5. The higher the <i>GAAP ETRVol</i> is, the higher the level of risk associated to the firm's tax position is. Columns 1 and 2 report at deviation of <i>GAAP ETR</i> from time t-1 to time t-5. The higher the <i>GAAP ETRVol</i> is, the higher the level of risk associated to the firm's tax position is. Columns 1 and 2 report at deviation of <i>GAAP ETR</i> from time t-1 to time t-5. The higher the <i>GAAP ETRVol</i> is, the higher the level of risk associated to the firm's tax position is. Columns 1 and 2 very at deviation of <i>GAAP ETR</i> for the results obtained by running the taxat dence, respectively. All the variables are winsorized at 1 and 99% to avoid the impact of outliers. All the models are estimated using standard errors clustered by firms, industry standard errors in parenthese. ***** $\rho < 0.01, **** \rho < 0.05, *** \rho < 0.1$

reported) and replicate the analysis. The results remain consistent with those in Columns 1 and 2. That is, our tax planning proxies are statistically significant in the expected direction after including life cycle stages.

Columns 5 and 6 show the models with the interaction term of tax planning and life cycle. The interaction variable measures the difference between the effect of tax planning on a group of firms in the same life cycle and the main effect (Wooldridge, 2010). Column 5 shows that the main effect of GAAPETR does not differ in growing and mature firms, but it differs in the introductory and declining firms. In the case of the latter firms, the coefficient of the interaction terms is positive and significant, signaling that introductory and declining firms with higher GAAP ETR (low tax avoidance) are more likely to default. Column 6 shows consistent and complementary findings. When GAAP ETRVol interacts with the life cycle. the results show that, in growing and mature firms, the effect of tax volatility on financial default is significantly higher than the average effect in the full sample, which is measured by the coefficient of GAAPETRVol (1.340) (Wooldridge, 2010). In contrast, the interaction of tax volatility and financial default has a negative and non-significant effect in introductory and declining firms, indicating that tax volatility does not contribute to a higher financial default in these firms. In this case, the interaction indicates that introductory and declining firms do not contribute to the average tax volatility effect on financial default in the full sample, which is measured by the coefficient of GAAP ETRVol (1.340) (Wooldridge, 2010).

These results are economically significant. Using hazard ratios, we show that, *ceteris paribus*, a 1% reduction in ETRs lowers the probability of default by 3.51 and 6.65% in introductory ($e^{(0.705-(-0.551))} = 3.51$) and declining firms ($e^{(1.344-(-0.551))} = 6.65$) [8], respectively. A 1% increase in the tax rate volatility increases the default likelihood by 1.92 and 2.36% in growing ($e^{(1.996-1.340)} = 1.92$) and mature firms ($e^{(2.200-1.340)} = 2.36$), respectively.

These findings support our hypotheses. Introductory and declining firms benefit from tax planning with a lower likelihood of default, confirming HP1 and HP4. Meanwhile, tax-planning growing and mature firms are more likely to default, confirming HP2 and HP3. We replicated the analysis using the logit model and obtained consistent results (available in a supplementary file upon request).

To ensure robust findings, we follow Hasan et al. (2017) and run a life cycle-wise analysis. Instead of including life cycle dummy variables together, we included each one by one. We then interacted each life cycle stage with the tax planning proxies. Table 4 shows the results for the hazard model and reports only the coefficients of tax planning and the interaction terms of tax planning and life cycle stages. Across all models in Table 4, the control variables are significantly associated with *Default* in the expected direction (not shown). Columns 1 and 2 in Table 4 report the results for the introduction stage. In Column 1, the GAAP ETR is significantly and negatively associated with Default, signifying that tax-avoiding firms are more likely to default. The coefficient of the interaction term of tax planning and the introduction stage is positive and highly significant, showing that introductory firms with higher ETRs are more likely to default. Hence, unlike other firms, tax-avoiding introductory firms benefit from tax planning with a lower likelihood of defaulting. Column 2 shows a consistent and complementary result; GAAP ETRVol is positively and significantly associated with *Default*, while the interaction of *GAAPETRVol* and the introduction stage is significantly and negatively associated. This finding indicates that introductory firms with higher tax volatility are much less likely to default than other firms in the sample. These findings provide empirical support for HP1, showing that tax-avoiding introductory firms are less likely to default. Columns 3 and 4 show that growing firms do not display a behavior different from the average behavior of the full sample, as indicated by the coefficients of the tax planning measures. The coefficients of the interaction of GAAPETR and GAAPETRVol with the life cycle stage are insignificant and have the same direction as that of the

332

MD

61.13

Hazard Model (life cycle-wise analysis)				Ċ	De	fault		L	
	Pred. Sign	(1)	ucnon (2)	(3) (3)	<i>wtn</i> (4)	(5) MI	ature (6)	T (J)	ecune (8)
Tax avoidance as GAAP ETR GAAP ETR _{itel}	(-)	-0.831*** (0.205)		-0.571**** (0.200)		-0.491**** (0.187)		-0.769*** (0.192)	
Tax avoidance as GAAP ETR volatility GAAP ETRVol _{4,(+1,45)}	(+)		2.647**** (0.450)		2.397*** (0.456)		1.973*** (0.448)		2.683**** (0.432)
$Life\ excle imes Tax\ avoidance$ $Introduction imes GAAP\ ETR_{4,1}$	(+)	1.066***							
$Growth imes GAAP ETR_{i,t-1}$	(-)	(0.309)		-0.363					
$Mature imes GAAP ETR_{i,t-1}$	(-)			(0.247)		-0.555**			
Decline $ imes$ GAAP ETR $_{ m it-1}$	(+)					(072.0)		1.737*** (0.503)	
$Life \ cycle imes Tax \ avoidance \ Introduction imes GAAP \ ETR Vol_{i}, (e.1,5)$	(-)		-2.012^{***}						
$Growth imes GAAP ETRVol_{i(t:1,t:5)}$	(+)		(0.724)		0.249				
$Mature imes GAAP ETRVol_{i,(t-1 t.5)}$	(+)				(060:0)		1.166*		
$Decline imes GAAP ETRVol_{i,(t-1, t-5)}$	(-)						(0.608)		-2.684***
Observations Stage-specific life cycle dumny Model controls Year fixed effects Industry fixed effects Standard error dustered by		90,790 YES YES YES FIRM	90,790 YES YES YES FIRM	90,790 YES YES YES FIRM	90,790 YES YES YES FIRM	90,790 YES YES YES YES FIRM	90,790 YES YES YES YES	90,790 YES YES YES YES FIRM	9.790 YES YES YES YES FIRM
Note(s): This table represents the Chapter 11 (restructuring) or Chaj Columns 3 and 4 report the results control variables included in the m standard errors clustered by firms Source(s): Created by authors	e estimated coeff pter 7 (liquidati for the growth s todel are lagged a s, industry and	icients obtainee on) at time t, fc tage. Columns at 1 year. All the year fixed effe	d by regressing or each stage of 5 and 6 report th 2 variables are w cts. Robust stan	the hazard moo the corporate e results for the vinsorized at 1s idard errors in	lel with a deper life cycle. Colu e mature stage. nd 99% to avo parentheses. **	ndent binary v mns 1 and 2 r Columns 7 and id the impact o ** $p < 0.01, **$	ariable (" <i>Default</i> eport the results 18 report the res f outliers. All the b < 0.05, * $p < 0$	") identifying s for the intro ults for the de e models are es .1	firms entering duction stage. cline stage. All stimated using
Table 4 Main life cycle-wis analysis: hazard mode								333	Tax planning and financia defaul

coefficients of the tax planning measures. This indicates that tax-planning growing firms are MD more likely to default. These findings support HP2. Similarly, Columns 5 and 6 show that the 61.13 coefficients of the interaction of GAAPETR and GAAPETRVol with the maturity stage are in the same direction as that of the coefficients of tax planning measures. This indicates that mature firms with lower effective rates and higher tax volatility are more likely to default. These findings support HP3. Finally, Column 7 shows that the effect of tax planning is reversed in declining firms. The coefficient of the interaction of GAAP ETR and the decline 334stage is positive and highly significant (1.737 and significant at the 1% level), while the coefficient of GAAP ETR is negative and significant. This result suggests that declining firms with lower ETRs (tax avoiders) are less likely to default. Column 8 shows that the prevailing effect of tax volatility on default is positive in the sample, but declining firms show a different behavior, as shown by the interaction term. Declining firms with higher tax volatility are also less likely to default, confirming that tax planning negatively affects financial default in these firms. These findings support HP4.

4.4 Robustness and endogeneity checks

To ensure robust findings, we ran our analyses using a different estimation model (the logit model) and a different measure of the dependent variable. In the main analyses, following previous studies (Altman, 1968; Beaver *et al.*, 2010), we measured financial default as a dummy variable that equaled one for each firm-year observation of a firm if the firm filed under Chapter 11 or Chapter 7. Otherwise, it equaled zero. To test our results' robustness, we measured financial default as a dummy variable that equals one for the firm-year observation in which a firm enters Chapters 11 or 7 and zero otherwise. We obtained results consistent with those reported above. These findings will be made available in a supplementary file upon request. We use another measure of the dependent variable and replicate the analysis. We use the Altman Z-score (Altman, 1968; Ali *et al.*, 2023) [9] as an alternative to financial default. The results remained consistent with our main findings (untabulated).

To address potential endogeneity bias from reverse causality, we employ an instrumental variable two-stage least squares (2SLS) approach that uses the fitted median value of tax planning as an instrumental variable for tax planning (Dhawan *et al.*, 2020; Hasan *et al.*, 2014). In the first stage, we regress the median value of tax planning on the life cycle and the interaction term of tax planning (*GAAP ETR*) and life cycle plus controls. We then used the first-stage regression to predict the median value of tax planning. This value (fitted median tax planning) is the instrumental variable (IV). We use this instrumental variable as an additional explanatory variable in our second-stage regression, which we performed by running the main pooled regression and replacing tax planning (*GAAP ETR*) with the fitted value of median tax planning. Table 5 reports the results of the 2SLS analysis with the two alternative measures of the dependent variable. Overall, the results suggest that endogeneity from reverse causality does not bias the results.

We also address potential endogeneity bias from omitted variables related to past negative performance. We re-ran our analyses and added controls for performance in the last three years (Table 6).

Table 6 reports the results obtained when we included a control for three-years lagged performance in the main model, as opposed to the one-year lagged performance used in the main analyses. Panels A and B in Table 6 report the results of using *GAAPETR* and *GAAP ETRVol* as a proxy for tax avoidance, respectively. In both the cases, we analyze data life cycle-wise. Collectively, the results confirm that past negative performance does not confound our results. Additional analyses on past performance, available in a supplementary file upon request, confirm that our findings are robust to endogeneity concerns.

		Defe	ault	
		(1)	(2)	and financial
IV	(-)	-1.088*	-1.064	default
	× /	(0.635)	(1.263)	
Introduction \times IV	(+)	1.378*	2.983*	
		(0.823)	(1.601)	
$Growth \times IV$	(-)	-0.717	-0.776	335
		(0.690)	(1.619)	
Mature imes IV	(-)	-0.818	-0.924	
		(0.701)	(1.571)	
Decline \times IV	(+)	2.595**	4.136*	
		(1.165)	(2.158)	
Observations		90,790	90,790	
Life cycle dummies		YES	YES	
Model controls		YES	YES	
Year fixed effects		YES	YES	
Industry fixed effects		YES	YES	
Std. error clustered by		FIRM	FIRM	

avoidance as an instrumental variable for tax avoidance (IV). Column 1 reports the result obtained by defining the default dependent dummy variable equal to 1 for each firm-year observations of a firm filing under either a Chapter 11 or Chapter 7 bankruptcy procedure, and 0 otherwise. Column 2 reports the results obtained by defining the default dependent dummy variable equal to 1 for the firm-year observations in which the firm enters a bankruptcy procedure, and 0 otherwise. All the variables are winsorized at 1 and 99% to avoid the impact of outliers. Robust standard errors in parentheses. ***p < 0.01, **p < 0.05, *p < 0.1 Source(s): Created by authors

Table 5.Addressingendogeneity from

potential reverse causality: instrumental variable (IV) approach

5. Discussion and contribution

5.1 Theoretical implications

Many studies on RBV have focused on firms' resource management across their life cycle and how managers obtain and use resources to improve performance and avoid financial defaults (Trahms *et al.*, 2013; Zahra, 2021; Islam and Fatema, 2023). Warnier *et al.* (2013) suggested that resource management research should not be limited to investigating strategic resources; it should be extended to a broader set of resources, such as cash resources, which are usually underexplored. Tax planning is a resource-generating activity for most businesses (Edwards *et al.*, 2016; Hanlon and Heitzman, 2010). However, RBV scholars have only recently acknowledged it as a resource-generating strategy (Hasan *et al.*, 2017), overlooked when investigating the firm's survival likelihood in distress conditions (Islam and Fatema, 2023).

A firm's life cycle stage influences two key conditions underlying the use of tax planning: a) *ex ante* availability of resources and b) resource management (Sirmon *et al.*, 2011; Trahms *et al.*, 2013; Zahra, 2021). Below, we discuss our results for each life cycle stage, referring to these two conditions and comparing our findings with that of prior literature.

Our results show that, in introductory firms, resource scarcity increases the propensity to obtain requisite cash resources from tax planning (Edwards *et al.*, 2016; Higgins *et al.*, 2015; Cerrato *et al.*, 2023). Simultaneously, owing to their small size and poor performance, tax planning by introductory firms has limited reputational risks and is unlikely to be strictly scrutinized by tax authorities (Sirmon *et al.*, 2011; Al-Hadi *et al.*, 2016). Additional resources obtained with limited risks reduce their default likelihood. Our findings suggest that tax planning complements other strategies identified by prior literature, which enhance the introductory firms' survival likelihood—such as reducing dividend payouts or changing top managers (Koh *et al.*, 2015; Trahms *et al.*, 2013).

MD 61,13		(1)	(2)	Default (3)	(4)	(5)
	PANEL A: Results after the inclusion of	f 3-year long ri	ın performanc	e using GAAP	ETR	
	Omitted variable – Past negative perfor 1.laggedEBIT_TA	mance -3.189***	-1.938***	-2.176***	-2.029***	-1.962***
336	2.laggedEBIT_TA	(0.373) -0.460^{***} (0.172)	(0.282) -0.771^{***} (0.120)	(0.282) -0.812^{***} (0.142)	(0.281) -0.797^{***} (0.141)	(0.284) -0.798^{***} (0.141)
	3.lagged EBIT_TA	(0.172) -0.883^{***} (0.136)	(0.139) -0.824^{***} (0.114)	(0.142) -0.923^{***} (0.115)	(0.141) -0.863^{***} (0.114)	(0.141) -0.898^{***} (0.115)
	Life cycle × Tax avoidance as CAAPE	TD				
	Life cycle \times Tax avoidance as GAAP E GAAP ETR _{i,t-1}	-0.627^{***} (0.224)	-0.708*** (0.210)	-0.421** (0.205)	-0.347* (0.193)	-0.619*** (0.198)
	Life cycle \times Tax avoidance Introduction \times GAAP ETR _{i,t-1}		1.147*** (0.318)			
	$\textit{Growth} \times \textit{GAAP ETR}_{i,t-1}$		(0.010)	-0.360 (0.251)		
	$Mature \times GAAP ETR_{i,t-1}$			(0.201)	-0.558** (0.272)	
	$\textit{Decline} \times \textit{GAAPETR}_{i,t-1}$					1.617*** (0.499)
	Observations Stage-specific life cycle dummy Model controlo	90,790 NO VES	90,790 YES VES	90,790 YES VES	90,790 YES	90,790 YES
	Year fixed effects	YES	YES	YES	YES	YES
	Std. error clustered by	FIRM	FIRM	FIRM	FIRM	FIRM
	PANEL B: Results after the inclusion of	f three-year lon	g run þerform	ance using GA	AP ETRVol	
	Omitted variable – Past negative perfor	mance				
	1.laggedEBIT_TA	-2.815^{***} (0.143)	-2.511^{***} (0.143)	-2.784^{***} (0.142)	-2.666^{***} (0.142)	-2.626^{***} (0.143)
	2.laggedEBIT_TA	-0.554^{***}	-0.502^{***}	-0.553^{***}	-0.524^{***}	-0.556^{***}
		(0.131)	(0.131)	(0.131)	(0.131)	(0.133)
	3.lagged EBIT_TA	-1.091*** (0.139)	-1.017*** (0.136)	-1.096*** (0.138)	-1.015*** (0.138)	-1.088^{***} (0.137)
	Life cycle \times Tax avoidance as GAAP E GAAP ETRVol _{i,(t-1,t-5)}	ETR volatility 2.427*** (0.342)	2.593*** (0.366)	2.328*** (0.368)	1.844*** (0.365)	2.633*** (0.347)
	Life cycle \times Tax avoidance Introduction \times GAAP ETRVol _{i,(t-1,t-5)}		-1.823***			
	$Growth imes GAAP ETRVol_{i,(t-1,t-5)}$		(0.579)	0.354		
T 11 <i>C</i>	$Mature imes GAAP ETRVol_{i,(t-1,t-5)}$			(0.489)	1.465***	
Addressing Addressing	$\textit{Decline} \times \textit{GAAP ETRVol}_{i,(t\text{-}1,t\text{-}5)}$				(0.484)	-2.935*** (0 9.30)
omitted variable bias: past negative	Observations	90,790	90,790	90,790	90,790	90,790
performance						(continued)

	(1)	(2)	Default (3)	(4)	(5)	Tax planning and financial
Stage-specific life cycle dummy	NO	YES	YES	YES	YES	default
Model controls	YES	YES	YES	YES	YES	
Year fixed effects	YES	YES	YES	YES	YES	
Industry fixed effects	YES	YES	YES	YES	YES	
Std. error clustered by	FIRM	FIRM	FIRM	FIRM	FIRM	337
Note(s): Table 6 Panel A displays the with <i>GAAP ETR</i> as independent variables and including a g9% to avoid the impact of outliers.	e coefficients of riable and inclu running the lif control for past Robust standard	otained by run iding a contro ie cycle-wise h performances l errors in pare	ning the life cy l for past perfe azard regressi s. All the varial entheses. *** <i>p</i>	cle-wise hazar ormances. Tab on with <i>GAA</i> oles are winsor < 0.01. ** <i>b</i> <	d regressions ble 6 Panel B P ETRVol as fized at 1 and 0.05, * $p < 0.1$	
Source(s): Created by authors	tobubt blundar	i errore in pui	p	0101, p	0100, p 011	Table 6.

Similar to introductory firms, declining firms must cope with resource scarcity (Koh *et al.*, 2015). Prior literature emphasizes how declining firms try to boost internal efficiency and gain support from key stakeholders (Koh *et al.*, 2015). We find that cash resources from tax planning are likely to be used for restructuring and fulfilling financial obligations as opposed to being utilized for lenders, suppliers and employees (Dickinson, 2011). Considering this, our findings suggest that tax planning can supplement divesting and restructuring activities that help avoid financial default (Zahra, 2021; Hasan *et al.*, 2017; Morrow *et al.*, 2007).

Our research demonstrates that growth firms engaging in aggressive tax planning are more likely to default. Since growing firms have strong operating cash flows and can raise cash resources through shareholders and lenders (Dickinson, 2011), tax planning generates slack resources, which weakens managerial discipline and induces inefficient usage of resources (Lee and Wu, 2016; George, 2005; Nohria and Gulati, 1996). In other words, unnecessary tax planning harms efficient resource management and may damage existing strategic resources (e.g. reputation) (Carnes *et al.*, 2017; Nohria and Gulati, 1996). Indeed, stakeholders perceive tax planning as a risky activity for growth firms; for instance, they penalize firms with a higher cost of debt or by limiting their supplies (Cerrato *et al.*, 2023; Hasan *et al.*, 2017; Flanagan and O'Shaughnessy, 2005).

Tax planning also increases the default risk for mature firms. Prior literature findings suggest that mature firms have abundant operating cash flows and poor investment opportunities (Faff *et al.*, 2016; Carnes *et al.*, 2017). Tax planning generates slack resources, which are likely used for managerial wealth extraction and/or to maintain ongoing negative net present value projects (Desai *et al.*, 2007). As in the case of growth firms, in a crucial stage for the firm's survival, unnecessary aggressive tax planning hinders efficient resource management and significantly damages the firm's reputation (Carnes *et al.*, 2017; Nohria and Gulati, 1996). Damages to reputation from tax planning result in an increased cost of debt and equity and a higher default risk (Hasan *et al.*, 2017; Duong *et al.*, 2022).

Collectively, our findings enhance the understanding of resource management in different life cycle stages of firms (Trahms *et al.*, 2013; Carnes *et al.*, 2017; Zahra, 2021). They contribute to prior literature with evidence that cash savings from tax planning—an overlooked non-VRIO resource—are managed and allocated in different ways in different life cycle stages, having a varied impact of default likelihood.

Our research can complement prior studies on slack resources, with evidence on cash resources. Existing research suggests that the influence of slack resources on performance or innovation depends on several factors, including the industry (Daniel *et al.*, 2004), type of firm (public or private; George, 2005), type of slack (absorbed or unabsorbed; Lee and Wu, 2016)

MD 61,13
 and the degree of managerial discretion (Magerakis, 2022). Nohria and Gulati (1996) demonstrated that both excessive or insufficient slack resources can hinder innovation. George (2005) demonstrated that high levels of slack resources hamper the performance of private firms but not of public firms. Lee and Wu (2016) found that slack resources have a curvilinear effect on the positive association between R&D expenses and firm performance. This study complements this literature by showing that the effect of slack resources on the likelihood of financial default depends on the nature of the slack and the life cycle stage. Excess unabsorbed slack cash resources resulting from tax planning can either increase or decrease the likelihood of default, depending on the life cycle stage.

The finding that tax planning and financial default are positively associated in growing and mature firms has another theoretical interpretation based on agency theory (Jensen and Meckling, 1976; Jensen, 1986). High levels of slack cash resources exacerbate agency conflicts between shareholders and managers (Jensen, 1986). Thus, firms that use tax planning to generate cash slack suffer from agency conflicts, causing non-optimal investment decisions, managerial wealth extraction and reputational risks (Balakrishnan *et al.*, 2019; Desai *et al.*, 2007; Dhawan *et al.*, 2020; Noga and Schnader, 2013; Shevlin *et al.*, 2020). Overall, our findings suggest that, regarding tax planning, cash slack has implications for both agency costs and the optimal management of firm resources at certain life cycle stages. Thus, RBV and agency theory may be complementary in explaining why tax-aggressive growing and mature firms are more likely to default. Our study supports the notion that studying the conceptual and empirical linkage between the agency theory and RBV can help overcome the problems related to "the unambiguous identification of appropriate resources, specification of testable hypotheses, and construction of suitable samples" (Lockett and Thompson, 2001, p. 743). It identifies cash as a key resource and provides testable hypotheses.

Finally, our results have relevant implications for research on corporate tax planning. Prior literature suggests that tax-planning activities are heterogeneous across a firm's life cycle (Cooper and Nguyen, 2020; Hasan *et al.*, 2017). Compared to Hasan *et al.*'s (2017) findings, we find that a firm's life cycle stage needs to be considered to not only understand why certain firms avoid more taxes than others but also gain insight into why only certain firms appear to benefit from tax planning.

5.2 Practical implications

Our findings inform entrepreneurs, managers and other practitioners interested in corporate tax planning. They underline the importance for managers to consider the costs and benefits associated with using tax planning as a resource-generating strategy in different life cycle stages. Scholes *et al.* (2014) posited that effective tax planning strategies should be the result of considering "all the parties," "all the costs," and "all the taxes" involved in decision-making. The tax benefits and non-tax costs (e.g. reputational and litigation risks) associated with tax planning likely change throughout a firm's life cycle (Hasan *et al.*, 2017). Therefore, managers should consider the firm's life cycle stage in determining tax planning strategies and balance the opportunity to create new resources with the need to preserve existing ones, with the firm's reputation being the most important.

This study is also of interest to managers assessing the costs and benefits of internal versus external financing (Cerrato *et al.*, 2023). Tax planning is a useful internal financing source with costs and benefits varying across life cycle stages and affecting the selection of internal versus external sources.

Our results also inform managers of the stakeholders' perceptions of tax planning. Tax planning is a powerful resource-generating strategy but also poses costs and risks for firms, such as tax audits, reputational risks and managerial opportunism (Jacob *et al.*, 2021). Investors, lenders, public authorities, employees and other stakeholders may not expect

aggressive tax planning from healthy and risk-adverse growing and mature firms. They could penalize such firms with a higher cost of debt, a higher cost of capital or reputational costs (Dhawan *et al.*, 2020; Hasan *et al.*, 2017). Thus, our results recommend managers to carefully consider the reactions of these stakeholders in deciding to plan taxes in different life cycle stages.

Finally, our study suggests to managers that tax planning may complement other strategies aimed at ensuring firm's survival. It also provides a quantitative estimate of the magnitude of the tax-planning effect on the firm's survival, holding great significance for managers and stakeholders (Mota *et al.*, 2022; Islam and Fatema, 2023).

5.3 Limitations and future research directions

This study has some limitations. First, it does not investigate firm-specific ownership structures and internal governance mechanisms, which may affect tax planning. In our research, we control for firm-level fixed effects and consider the firm's life cycle stage, which is related to the degree of development of internal governance structures (i.e. less developed in the introductory stage and more developed in later stages). However, some heterogeneity among firms within the same life cycle stage may prevail. Second, although this study uses robust measures like the GAAP ETR and cash ETR, there are other measures for tax planning (e.g. income shifting or cross-country transfer pricing) that may have varied effects across life cycle stages.

Future research could investigate whether ownership structures affect the relationship between tax planning and default, as moderated by life cycle; for instance, it could focus on family firms or analyze the demographics of key owners. It could also study whether internal governance mechanisms limit aggressive tax planning across the life cycle stages. Finally, future research could investigate the interaction between tax planning and other financing tools, like, e.g. working capital management or fintech (Liu *et al.*, 2023).

6. Conclusion

Drawing on the RBV, this study investigates the effect of tax planning on the likelihood of financial default at different life cycle stages. The results show that firms engaging in more aggressive tax planning are less likely to default in the introduction and decline stages. By contrast, tax-planning firms are more likely to default in the growth and maturity stages. The findings suggest that introductory and declining firms efficiently use cash resources from tax planning to meet their financial needs and acquire further resources that are useful for their survival. However, in growth and mature firms, tax aggressiveness produces unnecessary slack resources, weakens managerial discipline and increases reputational risks.

This study contributes to the RBV literature by answering the call for research on resource management at specific life cycle stages (Trahms *et al.*, 2013; Zahra, 2021). It shows that cash resources from tax planning are managed in distinctive ways in each life cycle stage, thereby having a varied impact on the likelihood of default. Our study sheds light on the underexplored non-VRIO cash resources, and highlights the potential linkages between the agency theory and RBV (Luu, 2023; Ali *et al.*, 2023).

The results of this research can be informative for managers, investors, lenders and other stakeholders, as it offers evidence on the tax planning benefits and costs across firms' life cycle stages.

Notes

1. Tax planning is the reduction of explicit taxes. It can include legal tax-reduction activities and aggressive tax-sheltering activities, which are likely challenged in a tax audit (Hanlon and Heitzman,

Tax planning and financial default

2010). The authors clarify that "tax avoidance represents a continuum of tax planning strategies where something like municipal bond investments is at one end (lower explicit tax, perfectly legal)
end" (Hanlon and Heitzman, 2010). In this study, tax planning is considered synonymous with tax
avoidance (see also Cooper and Nguyen, 2020).

- 2. Sirmon *et al.* (2007) defined resource management as "the comprehensive process of structuring, bundling and leveraging a firm's resources to creating value for customers and competitive advantages for the firm."
- 3. The literature review presented in this section is based on a systematic literature review for identifying and critically appraising relevant research (Snyder, 2019). The algorithmic tool used to search in relevant databases is available in a supplemental file to this paper, to ensure transparency and reproducibility (Davis *et al.*, 2014).
- 4. For example, Carnes *et al.* (2017) investigated a sample of growing and mature firms and found that firms manage their resource portfolio and capabilities to develop innovation based on the firm's life-cycle stage. Tariq *et al.* (2020) showed that, compared to growing firms, mature firms devote more financial and non-financial internal resources to undertake environmental investments.
- 5. Slack is a pool of resources over the minimum required for a given level of organizational output (Lee and Wu, 2016; Singh, 1986). Cash is a desirable slack for managers because it represents an unabsorbed available slack and is highly flexible and easily redeployed (Lee and Wu, 2016).
- 6. Chapter 11 of the US Bankruptcy Code is a court-supervised reorganization procedure aimed at reducing financial distress and allowing firms to continue to operate as a going concern. Chapter 7 of the US Bankruptcy Code is a liquidation procedure aimed at satisfying creditors' claims.
- 7. OANCF, IVNCF and FINCF is the operating cash flow from operating, investing and financing activities, respectively.
- 8. See Table 4.

MD 61.13

340

9. The authors generate a Z-score variable by using parameters from Altman (1968). Z-score = 1.2*(WC_TA) + 1.4*(RE_TA) +3.3*(Ebit_TA) +0.6(MV_TL) +0.999*(S_TA).

References

- Al-Hadi, A., Chatterjee, B., Yaftian, A., Taylor, G. and Monzur Hasan, M. (2019), "Corporate social responsibility performance, financial distress and firm life cycle: evidence from Australia", *Accounting and Finance*, Vol. 59 No. 2, pp. 961-989, doi: 10.1111/acfi.12277.
- Ali, S., Rehman, R.u., Aslam, S., Khan, I. and Murtaza, G. (2023), "Does board diversity reduce the likelihood of financial distress in the presence of a powerful Chinese CEO?", *Management Decision*, Vol. 61 No. 6, pp. 1798-1815, doi: 10.1108/MD-01-2022-0007.
- Altman, E.I. (1968), "Financial ratios, discriminant analysis and the prediction of corporate bankruptcy", *The Journal of Finance*, Vol. 23 No. 4, pp. 589-609, doi: 10.1111/j.1540-6261.1968.tb00843.x.
- Altman, E.I., Hotchkiss, E. and Wang, W. (2019), Corporate Financial Distress, Restructuring, and Bankruptcy: Analyze Leveraged Finance, Distressed Debt, and Bankruptcy, 4th ed., Jhon Wiley & Sons., Hoboken, NJ.
- Al-Hadi, A., Hasan, M.M. and Habib, A. (2016), "Risk committee, firm life cycle, and market risk disclosures", *Corporate Governance: An International Review*, Vol. 24 No. 2, pp. 145-170, doi: 10. 1111/corg.12115.
- Anandarajan, A., Chiang, S. and Lee, P. (2010), "R&D tax credit and operating performance: implications for managers", *Management Decision*, Vol. 48 No. 8, pp. 1198-1211, doi: 10.1108/ 00251741011076744.
- Andersén, J. (2011), "Strategic resources and firm performance", Management Decision, Vol. 49 No. 1, pp. 87-98, doi: 10.1108/00251741111094455.

Anthony, J.H. and Ramesh, K. (1992), "Association between accounting performance measures and stock prices: a test of the life cycle hypothesis", *Journal of Accounting and Economics*, Vol. 15 Nos 2-3, pp. 203-227.

- Badertscher, B.A., Katz, S.P. and Rego, S.O. (2013), "The separation of ownership and control and corporate tax avoidance", *Journal of Accounting and Economics*, Vol. 56 Nos 2-3, pp. 228-250, doi: 10.1016/j.jacceco.2013.08.005.
- Balakrishnan, K., Blouin, J.L. and Guay, W.R. (2019), "Tax aggressiveness and corporate transparency", *The Accounting Review*, Vol. 94 No. 1, pp. 45-69, doi: 10.2308/accr-52130.
- Bandopadhyaya, A. (1994), "An estimation of the hazard rate of firms under Chapter 11 protection", *The Review of Economics and Statistics*, Vol. 76 No. 2, pp. 346-350, doi: 10.2307/2109890.
- Barney, J. (1991), "Firm resources and sustained competitive advantage", Journal of Management, Vol. 17 No. 1, pp. 99-120, doi: 10.1177/014920639101700108.
- Beaver, W.H., Correia, M. and McNichols, M.F. (2010), "Financial statement analysis and the prediction of financial distress", *Foundations and Trends in Accounting*, Vol. 5 No. 2, pp. 99-173, doi: 10. 1561/1400000018.
- Biscotti, A.M., Mafrolla, E., Giudice, M.D. and D'Amico, E. (2018), "CEO turnover and the new leader propensity to open innovation: agency-resource dependence view and social identity perspective", *Management Decision*, Vol. 56 No. 6, pp. 1348-1364, doi: 10.1108/MD-04-2017-0392.
- Blouin, J. (2014), "Defining and measuring tax planning aggressiveness", National Tax Journal, Vol. 67 No. 4, pp. 875-899, doi: 10.17310/ntj.2014.4.06.
- Boyd, B.K., Bergh, D.D. and Ketchen, D.J. Jr (2010), "Reconsidering the reputation—performance relationship: a resource-based view", *Journal of Management*, Vol. 36 No. 3, pp. 588-609, doi: 10. 1177/0149206308328507.
- Campbell, J.Y., Hilscher, J. and Szilagyi, J. (2008), "In search of distress risk", *The Journal of Finance*, Vol. 63 No. 6, pp. 2899-2939, doi: 10.1111/j.1540-6261.2008.01416.x.
- Carnes, C.M., Chirico, F., Hitt, M.A., Huh, D.W. and Pisano, V. (2017), "Resource orchestration for innovation: structuring and bundling resources in growth-and maturity-stage firms", *Long Range Planning*, Vol. 50 No. 4, pp. 472-486, doi: 10.1016/j.lrp.2016.07.003.
- Cerrato, D., La Rocca, M. and Alessandri, T. (2023), "When does unrelated diversification increase performance? The effects of financial context and contingencies", *Management Decision*, Vol. 61 No. 13, pp. 1-25, doi: 10.1108/MD-02-2022-0225.
- Cooper, M. and Nguyen, Q.T. (2020), "Multinational enterprises and corporate tax planning: a review of literature and suggestions for a future research agenda", *International Business Review*, Vol. 29 No. 3, 101692, doi: 10.1016/j.ibusrev.2020.101692.
- Cuypers, I.R., Koh, P.S. and Wang, H. (2016), "Sincerity in corporate philanthropy, stakeholder perceptions and firm value", Organization Science, Vol. 27 No. 1, pp. 173-188, doi: 10.1287/orsc. 2015.1030.
- Daniel, F., Lohrke, F.T., Fornaciari, C.J. and Turner, R.A., Jr (2004), "Slack resources and firm performance: a meta-analysis", *Journal of Business Research*, Vol. 57 No. 6, pp. 565-574, doi: 10. 1016/S0148-2963(02)00439-3.
- Davis, J., Mengersen, K., Bennett, S. and Mazerolle, L. (2014), "Viewing systematic reviews and metaanalysis in social research through different lenses", *SpringerPlus*, Vol. 3, p. 511, doi: 10.1186/ 2193-1801-3-511.
- DeAngelo, H., DeAngelo, L. and Stulz, R.M. (2006), "Dividend policy and the earned/contributed capital mix: a test of the life-cycle theory x", *Journal of Financial Economics*, Vol. 81 No. 2, pp. 227-254, doi: 10.1016/j.jfineco.2005.07.005.
- Desai, M., Dyck, A. and Zingales, L. (2007), "Theft and taxes☆", Journal of Financial Economics, Vol. 84 No. 3, pp. 591-623, doi: 10.1016/j.jfineco.2006.05.005.

and financial default

Tax planning

MD 61,13	Dhawan, A., Ma, L. and Kim, M.H. (2020), "Effect of corporate tax avoidance activities on firm bankruptcy risk", <i>Journal of Contemporary Accounting and Economics</i> , Vol. 16 No. 2, pp. 1-23, doi: 10.1016/j.jcae.2020.100187.
	Dickinson, V. (2011), "Cash flow patterns as a proxy for firm life cycle", <i>The Accounting Review</i> , Vol. 86 No. 6, pp. 1969-1994, doi: 10.2308/accr-10130.
342	Drake, K.D., Lusch, S.J. and Stekelberg, J. (2019), "Does tax risk affect investor valuation of tax avoidance?", <i>Journal of Accounting, Auditing and Finance</i> , Vol. 34 No. 1, pp. 151-176, doi: 10. 1177/0148558X17692674.
	Duong, H.K., Fasan, M. and Gotti, G. (2022), "Living up to your codes? Corporate codes of ethics and the cost of equity capital", <i>Management Decision</i> , Vol. 60 No. 13, pp. 1-24, doi: 10.1108/MD-11- 2020-1486.
	Easley, D. and O'Hara, M. (2004), "Information and the cost of capital", Journal of Finance, Vol. 59

- No. 4, pp. 1553-1583, doi: 10.1111/j.1540-6261.2004.00672.x.
- Edwards, A., Schwab, C. and Shevlin, T. (2016), "Financial constraints and cash tax savings", *The Accounting Review*, Vol. 91 No. 3, pp. 859-881, doi: 10.2308/accr-51282.
- Ehsani, M. and Osiyevskyy, O. (2023), "Firm failure and the exploration/exploitation dilemma: the role of firm life cycle", *Long Range Planning*, Vol. 56 No. 3, 102307, doi: 10.1016/j.lrp.2023. 102307.
- Esteve-Pérez, S. and Mañez-Castillejo, J.A. (2008), "The resource-based theory of the firm and firm survival", *Small Business Economics*, Vol. 30, pp. 231-249, doi: 10.1007/s11187-006-9011-4.
- Faff, R., Kwok, W.C., Podolski, E.J. and Wong, G. (2016), "Do corporate policies follow a life-cycle?", *Journal of Banking and Finance*, Vol. 69, pp. 95-107, doi: 10.1016/j.jbankfin.2016.04.009.
- Fama, E.F. and Jensen, M.C. (1983), "Separation of ownership and control", Journal of Law and Economics, Vol. 26 No. 2, pp. 301-325, doi: 10.1086/467037.
- Filatotchev, I., Toms, S. and Wright, M. (2006), "The firm's strategic dynamics and corporate governance life-cycle", *International Journal of Managerial Finance*, Vol. 2 No. 4, pp. 256-279, doi: 10.1108/17439130610705481.
- Flanagan, D.J. and O'Shaughnessy, K.C. (2005), "The effect of layoffs on firm reputation", Journal of Management, Vol. 31 No. 3, pp. 445-463, doi: 10.1177/0149206304272186.
- Franczak, J., Pidduck, R.J., Lanivich, S.E. and Tang, J. (2023), "Immersed in Coleman's bathtub: multilevel dynamics driving new venture survival in emerging markets", *Management Decision*, Vol. 61 No. 7, pp. 1857-1887, doi: 10.1108/MD-03-2022-0308.
- Gallemore, J., Maydew, E.L. and Thornock, J.R. (2014), "The reputational costs of tax avoidance", Contemporary Accounting Research, Vol. 31 No. 4, pp. 1103-1133, doi: 10.1111/1911-3846.12055.
- George, G. (2005), "Slack resources and the performance of privately held firms", Academy of Management Journal, Vol. 48 No. 4, pp. 661-676, doi: 10.5465/amj.2005.17843944.
- Guenther, D.A., Matsunaga, S.R. and Williams, B.M. (2017), "Is tax avoidance related to firm risk?", *The Accounting Review*, Vol. 92 No. 1, pp. 115-136, doi: 10.2308/accr-51408.
- Habib, A. and Hasan, M.M. (2019), "Corporate life cycle research in accounting, finance and corporate governance: a survey, and directions for future research", *International Review of Financial Analysis*, Vol. 61 C, pp. 188-201, doi: 10.1016/j.irfa.2018.12.004.
- Hanlon, M. and Heitzman, S. (2010), "A review of tax research", *Journal of Accounting and Economics*, Vol. 50 Nos 2-3, pp. 127-178, doi: 10.1016/j.jacceco.2010.09.002.
- Hardeck, I. and Hertl, R. (2014), "Consumer reactions to corporate tax strategies: effects on corporate reputation and purchasing behavior", *Journal of Business Ethics*, Vol. 123, pp. 309-326, doi: 10. 1007/s10551-013-1843-7.
- Hasan, M.M. and Cheung, A. (2018), "Organization capital and firm life cycle", Journal of Corporate Finance, Vol. 48, pp. 556-578, doi: 10.1016/j.jcorpfin.2017.12.003.

- Hasan, I., Hoi, C.-K.S., Wu, Q. and Zhang, H. (2014), "Beauty is in the eye of the beholder: the effect of corporate tax avoidance on the cost of bank loans", *Journal of Financial Economics*, Vol. 113 No. 1, pp. 109-130, doi: 10.1016/j.jfineco.2014.03.004.
- Hasan, M.M., Al-Hadi, A., Taylor, G. and Richardson, G. (2017), "Does a firm's life cycle explain its propensity to engage in corporate tax avoidance?", *European Accounting Review*, Vol. 26 No. 3, pp. 469-501, doi: 10.1080/09638180.2016.1194220.
- Higgins, D., Omer, T.C. and Phillips, J.D. (2015), "The influence of a firm's business strategy on its tax aggressiveness", *Contemporary Accounting Research*, Vol. 32 No. 2, pp. 674-702, doi: 10.1111/ 1911-3846.12087.
- Hsu, F.J. (2018), "Does corporate social responsibility extend firm life-cycles?", Management Decision, Vol. 56 No. 11, pp. 2408-2436, doi: 10.1108/MD-09-2017-0865.
- Islam, M.M. and Fatema, F. (2023), "Do business strategies affect firms' survival during the COVID-19 pandemic? A global perspective", *Management Decision*, Vol. 61 No. 3, pp. 861-885, doi: 10.1108/ MD-11-2021-1456.
- Jacob, M., Rohlfing-Bastian, A. and Sandner, K. (2021), "Why do not all firms engage in tax avoidance?", *Review of Managerial Science*, Vol. 15, pp. 459-495, doi: 10.1007/s11846-019-00346-3.
- Jensen, M.C. (1986), "Agency cost of free cash flow, corporate finance, and takeovers", *The American Economic Review*, Vol. 76 No. 2, pp. 323-329.
- Jensen, M.C. and Meckling, W.H. (1976), "Theory of the firm: managerial behavior, agency costs and ownership structure", *Journal of Financial Economics*, Vol. 3 No. 4, pp. 305-360.
- Josefy, M.A., Harrison, J.S., Sirmon, D.G. and Carnes, C. (2017), "Living and dying: synthesizing the literature on firm survival and failure across stages of development", *Academy of Management Annals*, Vol. 11 No. 2, pp. 770-799, doi: 10.5465/annals.2015.0148.
- Koh, S., Durand, R.B., Dai, L. and Chang, M. (2015), "Financial distress: lifecycle and corporate restructuring", *Journal of Corporate Finance*, Vol. 33, pp. 19-33, doi: 10.1016/j.jcorpfin.2015.04.004.
- Kücher, A., Mayr, S., Mitter, C., Duller, C. and Feldbauer-Durstmüller, B. (2020), "Firm age dynamics and causes of corporate bankruptcy: age dependent explanations for business failure", *Review* of *Managerial Science*, Vol. 14 No. 3, pp. 633-661, doi: 10.1007/s11846-018-0303-2.
- Lee, C.L. and Wu, H.C. (2016), "How do slack resources affect the relationship between R&D expenditures and firm performance?", *R&D Management*, Vol. 46(S3), pp. 958-978, doi: 10.1111/ radm.12141.
- Liu, F., Wang, Q., Zhang, Z., Fang, M. and Xiao, S.(S.) (2023), "Lean inventory, fintech and financing: interactive influences on Chinese SMEs", *Management Decision*, Vol. 61 No. 8, pp. 2302-2321, doi: 10.1108/MD-06-2022-0878.
- Lockett, A. and Thompson, S. (2001), "The resource-based view and economics", Journal of Management, Vol. 27 No. 6, pp. 723-754, doi: 10.1177/014920630102700608.
- Lussier, R.N. (1995), "A nonfinancial business success versus failure prediction model for young firms", Journal of Small Business Management, Vol. 33 No. 1, pp. 8-20.
- Luu, T.D. (2023), "Fostering strategic entrepreneurship of SMEs: the role of organisational change forces", Management Decision, Vol. 61 No. 3, pp. 695-719, doi: 10.1108/MD-08-2021-1024.
- Magerakis, E. (2022), "The importance of managerial discretion on managerial ability-firm cash holding nexus", *Management Decision*, Vol. 60 No. 12, pp. 3275-3303, doi: 10.1108/MD-07-2021-0991.
- Maiti, M., Krakovich, V., Shams, S.R. and Vukovic, D.B. (2020), "Resource-based model for small innovative enterprises", *Management Decision*, Vol. 58 No. 8, pp. 1525-1541, doi: 10.1108/MD-06-2019-0725.
- Medioli, A., Azzali, S. and Mazza, T. (2020), "Ownership-motivated income shifting: evidence from European Multinational Groups", *Management Decision*, Vol. 58 No. 12, pp. 2621-2637, doi: 10. 1108/MD-08-2019-1048.

and financial default

Tax planning

 Morrov, JL. Jr. Sirmon, D.G., Hitt, M.A. and Holconb, T.R. (2007), "Creating value in the face or declining performance: firm strategies and organizational recovery", <i>Strategic Management Journal</i>, Vol. 28 No. 3, pp. 271-283, doi: 10.1002/smj.579. Mota, R.d.O., Bueno, A., Gonella, Jd.SL., Ganga, G.MD., Godinho Filho, M. and Latan, H. (2022), "The effects of the COVID-19 crisis on starturg's performance: the role of resilience", <i>Management Decision</i>, Vol. 60 No. 12, pp. 3388-3415, doi: 10.1108/MD/07.2021.0998. Newbert, S.L. (2007), "Empirical research on the resource-based view of the firm: an assessment and suggestions for future research", <i>Strategic Management Journal</i>, Vol. 29 No. 3, pp. 121-146 doi: 10.1002/smj.573. Noga, T.I. and Schnader, A.L. (2013), "Book-tax differences as an indicator of financial distress" <i>Accounting Horizons</i>, Vol. 27 No. 3, pp. 469-489, doi: 10.2308/acch.50481. Nohria, N. and Gulati, R. (1996), "Is slack good or bad for innovation?", <i>Academy of Management Journal</i>, Vol. 39 No. 5, pp. 122-1384, doi: 10.0465/2569898. Paeleman, L., Vanacker, T. and Zahra, S.A. (2023), "Should we be conservative or aggressive? SME managers responses in a crisis and long term firm survival", <i>Journal of Management Studies</i> (Garly view) doi: 10.1111/jonn.12993. Penrose, E.T. (1959), <i>The Theory of the Growth of the Firm</i>, 2nd ed, Basi Blackwell, London. Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches" <i>Review of Financial Studies</i>, Vol. 22 No. 1, p. 435-480, doi: 10.1083/rishlnin50. Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", <i>Journal of Business Research</i>, Vol. 135, pp. 391-407, doi: 10.1016/j. jbusres.2021.06.064. Quim, R.E. and Cameron, K. (1983), "Organizati	MD 61.13	Miller, D. and Friesen, P.H. (1984), "A longitudinal study of the corporate life cycle", Management Science, Vol. 30 No. 10, pp. 1161-1183, doi: 10.1287/mnsc.30.10.1161.
 Mota, R.d.O., Bueno, A., Gonella, J.d.S.L., Garga, G.M.D., Godinho Filho, M. and Latan, H. (2022), "The effects of the COVID-19 crisis on startups" performance: the role of resilience", <i>Management. Decision</i>, Vol. 60 No. 12, pp. 3388-3415, doi: 10.1108/MD/97.2021-0398. Newbert, S.L. (2007), "Empirical research on the resource-based view of the firm: an assessment and suggestions for future research", <i>Strategic Management Journal</i>, Vol. 28 No. 2, pp. 121-146 doi: 10.1002/smj.573. Noga, T.J. and Schnader, A.L. (2013), "Book-tax differences as an indicator of financial distress" <i>Accounting Hiorizons</i>, Vol. 27 No. 3, pp. 469-489, doi: 10.2608/acch.50481. Nohria, N. and Gulati, R. (1996), "Is slack good or bad for innovation?", <i>Academy of Management. Journal</i>, Vol. 39 No. 5, pp. 1245-1264, doi: 10.5465/256098. Paeleman, L., Vanacker, T. and Zahra, S.A. (2023), "Should we be conservative or aggressive? SMF managers' responses in a crisis and long-term firm survival", <i>Journal of Management Studies</i> (Early view) doi: 10.1111/joms.12993. Penrose, E.T. (1959), <i>The Theory of the Growth of the Firm</i>, 2nd ed, Basil Blackwell, London. Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches" <i>Review of Financial Studies</i>, Vol. 22 No. 1, pp. 435-480, doi: 10.1036/rishnto53. Purkayastha, A., Karma, A., Sharma, S. and Bhadra, D. (2021), "Poard's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", <i>Journal of Business Research</i>, Vol. 135, pp. 391-407, doi: 10.1026/j.pburse.2021.06.064. Quinn, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", <i>Management Science</i>, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnse.29 1.23. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress o	01,10	Morrow, J.L. Jr, Sirmon, D.G., Hitt, M.A. and Holcomb, T.R. (2007), "Creating value in the face of declining performance: firm strategies and organizational recovery", <i>Strategic Management</i> <i>Journal</i> , Vol. 28 No. 3, pp. 271-283, doi: 10.1002/smj.579.
 Newbert, S.L. (2007). "Empirical research", Strategic Management Journal, Vol. 28 No. 2, pp. 121-146 doi: 10.1002/smj.573. Noga, T.J. and Schnader, A.L. (2013), "Book-tax differences as an indicator of financial distress" Accounting Horizons, Vol. 27 No. 3, pp. 469-489, doi: 10.2308/acch.50481. Nohria, N. and Gulati, R. (1996), "Is slack good or bad for innovation?", Academy of Management Journal, Vol. 39 No. 5, pp. 1245-1264, doi: 10.5465/256948. Paeleman, I., Vanacker, T. and Zahra, S.A. (2023), "Should we be conservative or aggressive? SME managers' responses in a crisis and long term firm survival", Journal of Management Studies (Early view) doi: 10.1111/joms.12993. Penrose, E.T. (1959), The Theory of the Growth of the Firm, 2nd ed., Basil Blackwell, London. Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches" <i>Review of Financial Studies</i>, Vol. 22 No. 1, pp. 435-480, doi: 10.1093/rfs/hlm053. Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", Journal of Business Research, Vol. 135, pp. 391-407, doi: 10.1016/j.jbusres.2021.06.064. Quinn, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", Management Science, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnsc.29 1.33. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", Economic Modelling Vol. 44 C, pp. 44-53, doi: 10.1016/j.conamod.2014.09.015. Súnchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", Management Decision, Vol. 55 No. 5, pp. 842-871 doi: 10.1108/MD-12-2015-0666. Scholes, M., Wolfson, M., Er	344	Mota, R.d.O., Bueno, A., Gonella, J.d.S.L., Ganga, G.M.D., Godinho Filho, M. and Latan, H. (2022), "The effects of the COVID-19 crisis on startups' performance: the role of resilience", <i>Management Decision</i> , Vol. 60 No. 12, pp. 3388-3415, doi: 10.1108/MD-07-2021-0998.
 Noga, T.J. and Schnader, A.L. (2013), "Book-tax differences as an indicator of financial distress" Accounting Horizons, Vol. 27 No. 3, pp. 469-489, doi: 10.2308/acch-50481. Nohria, N. and Gulati, R. (1996), "Is slack good or bad for innovation?", Academy of Managemen. Journal, Vol. 39 No. 5, pp. 1245-1264, doi: 10.5465/256998. Paeleman, I., Vanacker, T. and Zahra, S.A. (2023), "Should we be conservative or aggressive? SME managers' responses in a crisis and long-term firm survival", Journal of Management Studies (Early view) doi: 10.1111/joms.12993. Penrose, E.T. (1959), <i>The Theory of the Growth of the Firm</i>, 2nd ed., Basil Blackwell, London. Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches" <i>Review of Financial Studies</i>, Vol. 22 No. 1, pp. 453-480, doi: 10.1083/rishhn053. Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", Journal of Business Research, Vol. 135, pp. 391-407, doi: 10.1016/j jbusres.2021.06.064. Quim, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", Management Science, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnsc.29 1.33. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", <i>Economic Modelling</i> Vol. 44 C, pp. 44-53, doi: 10.1016/j.econnod.2014.09.015. Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i>, Vol. 55 No. 5, pp. 842-871, doi: 10.1108/MD.12-2015-0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes an Business Strategy: A Planning Approach</i>, 5th ed., Pearson		Newbert, S.L. (2007), "Empirical research on the resource-based view of the firm: an assessment and suggestions for future research", <i>Strategic Management Journal</i> , Vol. 28 No. 2, pp. 121-146, doi: 10.1002/smj.573.
 Nohria, N. and Gulati, R. (1996), "Is slack good or bad for innovation?", Academy of Management Journal, Vol. 39 No. 5, pp. 1245-1264, doi: 10.5465/256498. Paeleman, I., Vanacker, T. and Zahra, S.A. (2023), "Should we be conservative or aggressive? SME managers' responses in a crisis and long-term firm survival", Journal of Management Studies (Early view) doi: 10.1111/joms.12993. Penrose, E.T. (1959), <i>The Theory of the Growth of the Firm</i>, 2nd ed., Basil Blackwell, London. Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches" <i>Review of Financial Studies</i>, Vol. 22 No. 1, pp. 435-480, doi: 10.1093/rfs/hhn053. Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", Journal of Business Research, Vol. 135, pp. 391-407, doi: 10.1016/j.jbusres.2021.06.064. Quim, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", Management Science, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mmsc.29 1.33. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", Economic Modelling Vol. 44 C, pp. 44-53, doi: 10.1016/j.commod.2014.09.015. Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", Management Decision, Vol. 55 No. 5, pp. 842-871 doi: 10.1108/MD12-2015.0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), Taxes and Business Strategy: A Planning Approach, 5th ed., Pearson Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", Journal of the American Taxation Association, Vol. 42 No. 2, pp. 1171		Noga, T.J. and Schnader, A.L. (2013), "Book-tax differences as an indicator of financial distress", <i>Accounting Horizons</i> , Vol. 27 No. 3, pp. 469-489, doi: 10.2308/acch-50481.
 Paeleman, I., Vanacker, T. and Zahra, S.A. (2023), "Should we be conservative or aggressive? SME managers' responses in a crisis and long-term firm survival", <i>Journal of Management Studies</i> (Early view) doi: 10.1111/joms.12993. Penrose, E.T. (1959), <i>The Theory of the Growth of the Firm</i>, 2nd ed., Basil Blackwell, London. Petersen, M.A. (2009), "Estimating standard errors in finance pand edata sets: comparing approaches" <i>Retiew of Financial Studies</i>, Vol. 22 No. 1, pp. 435-480, doi: 10.1093/rfs/hhn053. Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", <i>Journal of Business Research</i>, Vol. 135, pp. 391-407, doi: 10.1016/j.jbusres.2021.06.064. Quinn, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", <i>Management Science</i>, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnsc.29 1.33. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", <i>Economic Modelling</i> Vol. 44 C, pp. 44-53, doi: 10.1016/j.econmod.2014.09.015. Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i>, Vol. 55 No. 5, pp. 842-871 doi: 10.1108/MD1-22015-0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes and Business Strategy: A Planning Approach</i>, 5th ed., Pearson Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax-52605. Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vo		Nohria, N. and Gulati, R. (1996), "Is slack good or bad for innovation?", Academy of Management Journal, Vol. 39 No. 5, pp. 1245-1264, doi: 10.5465/256998.
 Penrose, E.T. (1959), <i>The Theory of the Growth of the Firm</i>, 2nd ed., Basil Blackwell, London. Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches" <i>Review of Financial Studies</i>, Vol. 22 No. 1, pp. 435-449, doi: 10.1083/rfs/h1n053. Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", <i>Journal of Business Research</i>, Vol. 135, pp. 391-407, doi: 10.1016/j jbusres.2021.06.064. Quinn, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", <i>Management Science</i>, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnsc.29 1.33. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", <i>Economic Modelling</i> Vol. 44 C, pp. 44-53, doi: 10.1016/j.econmod.2014.09.015. Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i>, Vol. 55 No. 5, pp. 842-871 doi: 10.1108/MD-12-2015-0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes and Business Strategy: A Planning Approach</i>, 5th ed., Perason Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.208/atax.52605. Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vol. 74 No. 1, pp. 101-124, doi: 10.1086/20665. Simph, J.V. (1986), "Performance, slack, and risk taking in organizational decision making", <i>Academy of Management Journal</i>, Vol. 29 No. 3, pp. 562-588, doi: 10.5465/256224. <li< td=""><td></td><td>Paeleman, I., Vanacker, T. and Zahra, S.A. (2023), "Should we be conservative or aggressive? SME managers' responses in a crisis and long-term firm survival", <i>Journal of Management Studies</i>. (Early view) doi: 10.1111/joms.12993.</td></li<>		Paeleman, I., Vanacker, T. and Zahra, S.A. (2023), "Should we be conservative or aggressive? SME managers' responses in a crisis and long-term firm survival", <i>Journal of Management Studies</i> . (Early view) doi: 10.1111/joms.12993.
 Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches" <i>Review of Financial Studies</i>, Vol. 22 No. 1, pp. 435-480, doi: 10.1093/rfs/hhn053. Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", <i>Journal of Business Research</i>, Vol. 135, pp. 391-407, doi: 10.1016/j jbusres.2021.06.064. Quinn, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", <i>Management Science</i>, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnsc.29 1.33. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", <i>Economic Modelling</i> Vol. 44 C, pp. 44-53, doi: 10.1016/j.econnuod.2014.09.015. Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i>, Vol. 55 No. 5, pp. 842-871 doi: 10.1108/MD-12-2015-0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes ana Business Strategy: A Planning Approach</i>, 5th ed., Pearson Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax.52605. Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Management Journal</i>, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. 2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Journal</i>, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/25624. Sirmon, D.G., Hitt, M.A., Ireland, R.D		Penrose, E.T. (1959), The Theory of the Growth of the Firm, 2nd ed., Basil Blackwell, London.
 Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency-resource dependence perspective", <i>Journal of Business Research</i>, Vol. 135, pp. 391-407, doi: 10.1016/j jbusres.2021.06.064. Quim, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", <i>Management Science</i>, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnsc.29 1.33. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", <i>Economic Modelling</i> Vol. 44 C, pp. 44-53, doi: 10.1016/j.econmod.2014.09.015. Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i>, Vol. 55 No. 5, pp. 842-871. doi: 10.108/MD-12-2015-0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes and Business Strategy: A Planning Approach</i>, 5th ed., Pearson Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax.52605. Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vol. 74 No. 1, pp. 101-124, doi: 10.1086/209665. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Journal</i>, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i>, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310		Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches", <i>Review of Financial Studies</i> , Vol. 22 No. 1, pp. 435-480, doi: 10.1093/rfs/hhn053.
 Quinn, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness some preliminary evidence", <i>Management Science</i>, Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnsc.29 1.33. Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", <i>Economic Modelling</i> Vol. 44 C, pp. 44-53, doi: 10.1016/j.econmod.2014.09.015. Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i>, Vol. 55 No. 5, pp. 842-871 doi: 10.1108/MD-12-2015-0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes ana Business Strategy: A Planning Approach</i>, 5th ed., Pearson Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax-52605. Shunway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vol. 74 No. 1, pp. 101-124, doi: 10.1086/209665. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Review</i>, Vol. 32 No. 1, pp. 273-292, doi: 10.5465/amr.2007.23466005. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage breadth, depth, and life cycle effects", <i>Journal of Management</i>, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695. 		Purkayastha, A., Karna, A., Sharma, S. and Bhadra, D. (2021), "Board's human capital resource and internationalization of emerging market firms: toward an integrated agency–resource dependence perspective", <i>Journal of Business Research</i> , Vol. 135, pp. 391-407, doi: 10.1016/j. jbusres.2021.06.064.
 Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", <i>Economic Modelling</i> Vol. 44 C, pp. 44-53, doi: 10.1016/j.econmod.2014.09.015. Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i>, Vol. 55 No. 5, pp. 842-871 doi: 10.1108/MD-12-2015-0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes and Business Strategy: A Planning Approach</i>, 5th ed., Pearson Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax-52605. Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vol. 74 No. 1, pp. 101-124, doi: 10.1086/209665. Singh, J.V. (1986), "Performance, slack, and risk taking in organizational decision making", <i>Academy of Management Journal</i>, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Review</i>, Vol. 32 No. 1, pp. 273-292, doi: 10.5465/amr.2007.23466005. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i>, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695. 		Quinn, R.E. and Cameron, K. (1983), "Organizational life cycles and shifting criteria of effectiveness: some preliminary evidence", <i>Management Science</i> , Vol. 29 No. 1, pp. 33-51, doi: 10.1287/mnsc.29. 1.33.
 Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i>, Vol. 55 No. 5, pp. 842-871 doi: 10.1108/MD-12-2015-0566. Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes and Business Strategy: A Planning Approach</i>, 5th ed., Pearson Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax-52605. Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vol. 74 No. 1, pp. 101-124, doi: 10.1086/209665. Singh, J.V. (1986), "Performance, slack, and risk taking in organizational decision making", <i>Academy of Management Journal</i>, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Review</i>, Vol. 32 No. 1, pp. 273-292, doi: 10.5465/amr.2007.23466005. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i>, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695. 		Richardson, G., Taylor, G. and Lanis, R. (2015), "The impact of financial distress on corporate tax avoidance spanning the global financial crisis: evidence from Australia", <i>Economic Modelling</i> , Vol. 44 C, pp. 44-53, doi: 10.1016/j.econmod.2014.09.015.
 Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes and Business Strategy: A Planning Approach</i>, 5th ed., Pearson Publishing, London. Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax-52605. Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vol. 74 No. 1, pp. 101-124, doi: 10.1086/209665. Singh, J.V. (1986), "Performance, slack, and risk taking in organizational decision making", <i>Academy of Management Journal</i>, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Review</i>, Vol. 32 No. 1, pp. 273-292, doi: 10.5465/amr.2007.23466005. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i>, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695. 		Sánchez, L.P.C., Guerrero-Villegas, J. and Hurtado Gonzalez, J.M. (2017), "The influence of organizational factors on board roles", <i>Management Decision</i> , Vol. 55 No. 5, pp. 842-871, doi: 10.1108/MD-12-2015-0566.
 Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i>, Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax-52605. Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vol. 74 No. 1, pp. 101-124, doi: 10.1086/209665. Singh, J.V. (1986), "Performance, slack, and risk taking in organizational decision making", <i>Academy of Management Journal</i>, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Review</i>, Vol. 32 No. 1, pp. 273-292, doi: 10.5465/amr.2007.23466005. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i>, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695. 		Scholes, M., Wolfson, M., Erickson, M., Hanlon, M., Maydew, E. and Shevlin, T. (2014), <i>Taxes and Business Strategy: A Planning Approach</i> , 5th ed., Pearson Publishing, London.
 Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i>, Vol. 74 No. 1, pp. 101-124, doi: 10.1086/209665. Singh, J.V. (1986), "Performance, slack, and risk taking in organizational decision making", <i>Academy of Management Journal</i>, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Review</i>, Vol. 32 No. 1, pp. 273-292, doi: 10.5465/amr.2007.23466005. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i>, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695. 		Shevlin, T., Urcan, O. and Vasvari, F.P. (2020), "Corporate tax avoidance and debt costs", <i>Journal of the American Taxation Association</i> , Vol. 42 No. 2, pp. 117-143, doi: 10.2308/atax-52605.
 Singh, J.V. (1986), "Performance, slack, and risk taking in organizational decision making", Academy of Management Journal, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224. Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", Academy of Management Review, Vol. 32 No. 1, pp. 273-292, doi: 10.5465/anr.2007.23466005. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", Journal of Management, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695. 		Shumway, T. (2001), "Forecasting bankruptcy more accurately: a simple hazard model", <i>Journal of Business</i> , Vol. 74 No. 1, pp. 101-124, doi: 10.1086/209665.
 Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Review</i>, Vol. 32 No. 1, pp. 273-292, doi: 10.5465/amr.2007.23466005. Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i>, Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695. 		Singh, J.V. (1986), "Performance, slack, and risk taking in organizational decision making", <i>Academy</i> of Management Journal, Vol. 29 No. 3, pp. 562-585, doi: 10.5465/256224.
Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i> , Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695.		Sirmon, D.G., Hitt, M.A. and Ireland, R.D. (2007), "Managing firm resources in dynamic environments to create value: looking inside the black box", <i>Academy of Management Review</i> , Vol. 32 No. 1, pp. 273-292, doi: 10.5465/amr.2007.23466005.
		Sirmon, D.G., Hitt, M.A., Ireland, R.D. and Gilbert, B.A. (2011), "Resource orchestration to create competitive advantage: breadth, depth, and life cycle effects", <i>Journal of Management</i> , Vol. 37 No. 5, pp. 1390-1412, doi: 10.1177/0149206310385695.

- Snyder, H. (2019), "Literature review as a research methodology: an overview and guidelines", Journal of Business Research, Vol. 104, pp. 333-339, doi: 10.1016/j.jbusres.2019.07.039.
- Song, J., Wen, S., Ren, Q. and Zhang, L. (2022), "The choice and acquisition of external knowledge in startups: matching improvisation to strategic flexibility", *Management Decision*, Vol. 60 No. 11, pp. 2928-2951, doi: 10.1108/MD-04-2021-0467.
- Tariq, A., Badir, Y.F., Safdar, U., Tariq, W. and Badar, K. (2020), "Linking firms' life cycle, capabilities, and green innovation", *Journal of Manufacturing Technology Management*, Vol. 31 No. 2, pp. 284-305, doi: 10.1108/JMTM-08-2018-0257.
- Trahms, C.A., Ndofor, H.A. and Sirmon, D.G. (2013), "Organizational decline and turnaround: a review and agenda for future research", *Journal of Management*, Vol. 39 No. 5, pp. 1277-1307, doi: 10. 1177/0149206312471390.
- Vergara, R. (2010), "Taxation and private investment: evidence for Chile", *Applied Economics*, Vol. 42 No. 6, pp. 717-725, doi: 10.1080/00036840701720747.
- Wang, F., Xu, S., Sun, J. and Cullinan, C.P. (2020), "Corporate tax avoidance: a literature review and research agenda", *Journal of Economic Surveys*, Vol. 34 No. 4, pp. 793-811, doi: 10.1111/ joes.12347.
- Warnier, V., Weppe, X. and Lecocq, X. (2013), "Extending resource-based theory: considering strategic, ordinary and junk resources", *Management Decision*, Vol. 51 No. 7, pp. 1259-1379, doi: 10.1108/MD-05-2012-0392.
- Watson, L. (2015), "Corporate social responsibility, tax avoidance, and earnings performance", The Journal of the American Taxation Association, Vol. 37 No. 2, pp. 1-21, doi: 10.2308/atax-51022.
- Wernerfelt, B. (1984), "A resource-based view of the firm", Strategic Management Journal, Vol. 5 No. 2, pp. 171-180, doi: 10.1002/smj.4250050207.
- Wooldridge, J.M. (2010), Econometric Analysis of Cross Section and Panel Data, 2nd ed., MIT Press, Cambridge, MA.
- Xu, S. and Zheng, K. (2020), "Tax avoidance and asymmetric cost behavior", *Journal of Accounting, Auditing and Finance*, Vol. 35 No. 4, pp. 723-747, doi: 10.1177/0148558X18793757.
- Yang, L., Qin, H., Xia, W., Gan, Q., Li, L., Su, J. and Yu, X. (2021), "Resource slack, environmental management maturity and enterprise environmental protection investment: an enterprise life cycle adjustment perspective", *Journal of Cleaner Production*, Vol. 309, 127339, doi: 10.1016/j. jclepro.2021.127339.
- Zahra, S.A. (2021), "The resource-based view, resourcefulness, and resource management in startup firms: a proposed research agenda", *Journal of Management*, Vol. 47 No. 7, pp. 1841-1860, doi: 10.1177/01492063211018505.
- Zimmerman, J.L. (1983), "Taxes and firm size", Journal of Accounting and Economics, Vol. 5, pp. 119-149, doi: 10.1016/0165-4101(83)90008-3.
- Zona, F., Gomez-Mejia, L.R. and Withers, M.C. (2018), "Board interlocks and firm performance: toward a combined agency–resource dependence perspective", *Journal of Management*, Vol. 44 No. 2, pp. 589-618, doi: 10.1177/0149206315579512.

Further reading

- Brondolo, J. (2009), "Collecting taxes during an economic crisis: challenges and policy options", *IMF Staff Position Notes*, Vol. 2009 No. 17, doi: 10.5089/9781462339440.004.
- Xue, S., Chen, H. and Wu, J. (2023), "Performance feedback and firms' strategic actions: a crossnational meta-analysis", *Management Decision*, Vol. 61 No. 7, pp. 2201-2230.

(The Appendix follows overleaf)

default

345

Tax planning

and financial

Supplemental File Tax planning and financial default: role of corporate life cycle"

		Non-defaulted firms	Defaulted firms	Total
346	Introduction	6,596 (92.37%)	545 (7.63%)	7,141 (7.87%)
	Growth	<i>29,430</i> (98.50%)	419 (1.5%)	29,849 (32.87%)
	Mature	40,267 (98.64%)	554 (1.36%)	40,821 (44.96%)
	Shake-out	9,970 (96.8%)	<i>322</i> (3.12%)	10,292 (11.35%)
Table S1. Breakdown defaulted	Decline Total	<i>2,406</i> (89.54%) <i>88,669</i> (97.6%)	<i>281</i> (10.46%) <i>2,121</i> (2.3%)	<i>2,687</i> (2.95%) <i>90,790</i> (100%)
firm-year observation for each life cycle's stages	Note(s): This tak stage of the corpo Source(s): Creat	ble shows the defaulted, non-defau brate life cycle at time t ted by authors	ilted and the total number of firm-y	ear observations for each

MD 61,13

	Pred. Sign	(1)	(2)	Default (3)	(4)	(5)	(9)
voidance as GAAP ETR P ETR _{ite1}	(-)	-0.807^{***} (0.210)		-0.749^{***} (0.211)		-0.616* (0.373)	
avoidance as GAAP ETR volatility P ETR Vol _{it} (t-1, t-5)	(+)		3.159*** (0.368)		2.827*** (0.371)		1.727** (0.698)
ycle \times Tax avoidance duction \times GAAP ETR _{1, t-1}	(+)					0.642	
$th imes GAAP ETR_{i,t-1}$	(-)					(0.473) -0.450	
$we imes GAAP ETR_{ m i,t-1}$	(-)					(0.42b) -0.508	
$ne imes GAAP ETR_{ m it+1}$	(+)					(0.437) 1.292** (0.630)	
ycle \times Tax avoidance duction \times GAAP ETR Vol _{4(t-1, t-5)}	(-)						0.699
$th imes GAAP ETRVol_{i,(t-1, t-5)}$	(+)						(0.884) 2.033**
$re imes GAAPETRVol_{i,(t-1, t-5)}$	(+)						2.195*** 2.195***
$\iota e imes GAAP ETRVol_{i,(t-1, t-5)}$	(-)						(0.709) -1.238 (1.056)
							(continued)
Table S2. Main pooled analysis:						347	Tax planning and financial default

MD 61,13	(9)	90,790 YES YES YES YES YES FIRM 0.149 0.149 0.149 ug firms entering APETR variable, 2AAP ETR is, the candard deviation (variables used in cet value of equity e results obtained including the life e life cycle, (2) tax inpact of outliers. ses. **** $p < 0.01$,
348	(2)	90,790 YES YES YES YES YES YES FIRM 0.133 0.133 0.133 1.10 entifyi a 1.vear lagge GA 0. The higher the C 0. The higher the C 0. The higher the C 1. the logat model and sition is. The control sition is a set of the control sition is in the control set of the
	sfault (4)	90,790 YES YES YES YES YES AFRW 0.14800000000000000000000000000000000000
	(3) De	90,790 YES YES YES YES YES PIRM 0.133 lel with a depender aport time t-1 (Hanlor 1-year lagged GAA 1-year lagged GAA frisk associated wi frisk associated wi fris
	(2)	90,790 NO YES YES YES YES FIRM 0.135 ssing the logit moc oldimed by using a obtained by using a obtained by using a net the results obtain lance, respectively, ns, industry and y
	(1)	90,790 NO YES YES YES YES YES FIRM 0.121 obtained by regre cedures at time t. C e t-1 on the pre-tax neport the results of the PTR/ofis, th 7A). (2) retained as S_TAI . All control- tax avoidance. Col and Column 6 repo cycle and tax avoid clustered by firm
	Pred. Sign	7 mated coefficients 7 (liquet coefficients tax expense ant tim columns 2, 4 and 6 5. The higher the G for the sects (<i>WC</i> - les on total assets (<i>WC</i> - les on total assets (<i>WC</i> - les on total assets for ing our proxies for oidance. Column 5, veen corporate life.
Table S2.		servations se cycle dummies odel controls ar fixed effects an fixed effects andard errors clustered by eudo R2 tedo R2 t

<i>Logit Model</i> (life cycle-wise analysis)		Introduct	ion	Growth	Default	Mature		Decline	
	Pred. Sign	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Tax avoidance as (GAAP ETR _{it-1}	GAAP ETR (-)	-0.930*** (0.228)		-0.718^{***} (0.225)		-0.632*** (0.211)		-0.876*** (0.215)	
Tax avoidance as (GAAP ETRVol _{i,(t-1,t-5)}	3AAP ETR V((+)	olatility	3.334*** (0.390)		2.997*** (0.394)		2.615*** (0.394)		3.331*** (0.376)
Life cycle \times Tax an Introduction \times GAAP ETR _{it-1} Grouth \times GAAP ETR _{it-1} Mature \times GAAP ETR _{it-1} Decline \times GAAP ETR _{it-1}	<i>voidance</i> (+) (-) (+) (+)	1.004*** (0.360)		-0.290 (0.275)		-0.495* (0.298)		1.665*** (0.566)	
Life cycle \times Tax an httroduction \times GAAP ETRVol _i , ($^{(+1+5)}$ Growth \times GAAP ETRVol _i ($^{(+1+5)}$ Mature \times GAAP ETRVol _i ($^{(+1+5)}$ Decime \times GAAP ETRVol _i ($^{(+1+5)}$	oidance (-) (+) (-)		-2.374^{***} (0.654)		0.583 (0.510)		1.190** (0.515)		-2.806*** (0.940)
Table S3. Life cycle-wise analysis: logit model								349	Tax planning and financial default

MD 61,13	line	(8)	90,790 YES	YES YES YES	FIRM	g firms entering eport the results included in the ors clustered by
350	De	(2)	90,790 YES	YES YES YES	FIRM	<i>Default</i> ") identifyin Columns 3 and 41 All control variable using standard en using standard en
	ature	(9)	90,790 YES	YES YES YES	FIRM	inary variable ("L atroduction stage. he decline stage. A lels are estimated
	efault M	(2)	90,790 YES	YES YES YES	FIRM	th a dependent b he results for the in rt the results for the rt theresults for t titiers. All the mod 0.05, *p < 0.1
	Powth D	(4)	90,790 YES	YES YES YES	FIRM	he logit model wi s 1 and 2 report th imms 7 and 8 repo d the impact of ou $^{**}p < 0.01, \ ^{**}p <$
	6	(3)	90,790 YES	YES YES YES	FIRM	d by regressing t at time t. Column nature stage. Colu and 99% to avoit in parentheses. **
	oduction	(2)	90,790 YES	YES YES YES	FIRM	efficients obtaine ation) procedures he results for the 1 re winsorized at 1 : standard errors
	Intr	(1)	90,790 YES	YES YES YES	FIRM	the estimated co Chapter 7 (liquid as 5 and 6 report t All the variables a ed effects. Robust tors
Table S3.	Logit Model (life cycle-wise analysis) Pred.	Sign	Observations Stage-specific life	Cycle dummy Model controls Year fixed effects Industry fixed	enects Standard error clustered by	Note(s): This table reports Chapter 11 (restructuring) on for the growth stage. Column model are lagged at 1-year. <i>i</i> firms, industry and year fixu Source(s): Created by auth

	Pred. Sign	(1)	efault (2)	Tax planning and financial
<i>Tax avoidance as GAAP ETR GAAP ETR</i>	()	-0.784 (0.768)		default
Tax avoidance as GAAP ETR volatility GAAP ETRVol _{i,(t-1,t-5)}	(+)		0.963 (1.178)	351
Life cycle \times Tax avoidance Introduction \times GAAP ETR _{i,t-1}	(+)	1.628*		
$Growth \times GAAPETR_{i,t-1}$	(—)	(0.372) -0.287 (1.027)		
$Mature imes GAAP ETR_{i,t-1}$	()	(1.037) -0.225 (1.010)		
Decline \times GAAP ETR _{i,t-1}	(+)	(1.016) 2.203* (1.236)		
Life cycle \times Tax avoidance Introduction \times GAAP ETRVol _{i,(t-1,t-5)}	(—)		-1.556	
$Growth \times GAAP ETRVol_{i,(t-1,t-5)}$	(+)		(1.497) 1.114 (1.642)	
$Mature \times GAAP ETRVol_{i,(t-1,t-5)}$	(+)		(1.043) 3.926*** (1.490)	
$Decline imes GAAP ETRVol_{i,(t-1,t-5)}$	(—)		(1.480) -1.311 (1.706)	
Observations Life cycle dummies Model controls Year fixed effects Industry fixed effects Standard errors clustered by		90,790 YES YES YES YES FIRM	(1.796) 90,790 YES YES YES YES FIRM	

Note(s): This table reports the pooled estimated coefficients obtained by running the logit model and using an alternative definition of the dependent variable. In this table, the *Default* dependent variable is defined as a dummy variable equal to 1 for the firm-year observations in which the firm enters the Chapter 11 or Chapter 7 procedure, and 0 otherwise. Column 1 reports the results obtained by using the 1-year lagged *GAAPETR* as explanatory variable for tax avoidance. Column 2 reports the results obtained by using the 1-year lagged *GAAPETRVol* as explanatory variable for tax avoidance. Control variables used in the models are: (1) working capital on total assets (*WC_TA*), (2) retained earnings on total assets (*RE_TA*), (c) EBIT on total assets (*EBIT_TA*), (4) market value of equity on total liabilities (*MV_TL*) and (5) sales on total assets (*S_TA*). All control variables included in the model are lagged at 1-year. Both columns report the results obtained by running the logit model and including: (1) corporate life cycle, (2) tax avoidance are winsorized at 1 and 99% to avoid the impact of outliers. Robust standard errors in parentheses. ***p < 0.01, **p < 0.05, *p < 0.1 **Source(s):** Created by authors

Table S4.Robustness check:pooled analysis using
an alternative
definition of the
dependent variable

$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Table S5. Robustness check: life cycle-wise analysis using an alternative								352	MD 61,13
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $			L. true J				fault	(at 110)		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	đ	hed	INITOOXI	ucnon	5	rowth	-W	lature	De	scane
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	101	Sign	(1)	(2)	(3)	(4)	(2)	(9)	(1)	(8)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	pidance as GAAP ETR ETR _{i,t-1}		-0.822** (0.406)		-0.213 (0.400)		-0.148 (0.405)		-0.603 (0.372)	
$ \begin{array}{ccccc} k \times Tax a avoidance \\ eiton \times GAAP \\ \times GAAP ETR_{4,} & (-) & \underbrace{1.900^{4446}}_{(0.733)} & -0.817 \\ \times GAAP ETR_{4,} & (-) & \underbrace{0.733}_{(0.733)} & -0.817 \\ \times GAAP ETR_{4,} & (-) & \underbrace{0.733}_{(0.733)} & -0.817 \\ \times GAAP ETR_{4,} & (-) & \underbrace{0.733}_{(0.733)} & -0.823 \\ \times GAAP ETR_{4,} & (-) & \underbrace{3507^{444}}_{(1.022)} & -0.424 \\ (-) & \underbrace{3507^{444}}_{(1.1022)} & -0.424$	vidance as GAAP ETR ETRVol _{i,(t-1,t-5)}	volatility (+)		2.803*** (0.545)		2.272*** (0.515)		1.043* (0.547)		2.518*** (0.500)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	te × Tax avoidance ction × GAAP × GAAP ETR _{it} .	$(\hat{\pm})$ $(\hat{\pm})$	1.900*** (0.733)		-0.817					
$ \begin{array}{cccc} \times GAA ETR_{1,1} & (+) & (-) & ($	\times GAAP ETR _{i,t} .	(-)			(0.783)		-0.893			
$ \begin{array}{cccc} k < Tax avoidance \\ tion \times GAAP \\ (+) \\ \times GAAP \\ (+) \\ \times GAAP \\ (+) \\ (1.042) \\ \times GAAP \\ (+) \\ (-) \\ \times GAAP \\ (-) \\$	\times GAAP ETR _{it-1}	(+)					(80.1.0)		2.359** (1.055)	
$\begin{array}{ccccccc} & & & & & & & & & & & & & & & &$	$ \begin{array}{l} e \times Tax \ avoidance \\ ction \times GAAP \\ t_1 (e_{11,e_3}) \\ \times GAAP \\ t_{1(e_{11,e_3})} \end{array} $	$(-)$ (\pm) (\pm) $(-)$		-3.507*** (1.092)		-0.424 (1.235)		3.615*** (1.041)		ر محمد المحمد ا
(contin	tite1, t-5) titions becific life cycle		90,790 YES	90,790 YES	90,790 YES	90,790 YES	90,790 YES	90,790 YES	90,790 YES	00,790 YES
										(continued)

Introduction Growth Default	red. Sign (1) (2) (3) (4)	YES YES YES YES YES YES YES YES YES YES YES YES YES FIRM FIRM FIRM FIRM I	he coefficients for each stage of the corporate life cycle obtained by running 1 and 2 report the results for the introduction stage. Columns 3 and 4 report th olumns 7 and 8 report the results for the decline stage. All the variables are wins s , **** $p < 0.01$, *** $p < 0.05$, * $p < 0.1$		
Mature	(5) (6)	YES YES YES YES YES YES FIRM FIRM	the logit model and using he results for the growth str sorized at 1 and 99% to avo		
Decline	(7) (8)	YES YES YES YES YES YES FIRM FIRM	: an alternative definition of th age. Columns 5 and 6 report th aid the impact of outliers. Robus	_	and fin

ble S5.

353

nning ancial efault

MD			Defe	rult					
61,13		(1)	(2)	(3)	(4)				
	PANEL A: Results after the inclusion of 3-year long run performance using GAAP ETR (Logit Model)								
354	Omitted variable – Past negative perfor 1.laggedEBIT_TA 2.laggedEBIT_TA	mance	-3.296*** (0.355) -0.868*** (0.160)	-3.085^{***} (0.353) -0.849^{***} (0.160)	-3.050^{***} (0.356) -0.855^{***} (0.160)				
	3.lagged EBIT_TA	-0.966*** (0.130)	-1.067*** (0.130)	-1.002*** (0.130)	-1.036*** (0.130)				
	Life cycle \times Tax avoidance as GAAP E GAAP ETR _{i,t-1}	CTR -0.801*** (0.235)	-0.560** (0.232)	-0.485** (0.220)	-0.726*** (0.223)				
	Life cycle \times Tax avoidance Introduction \times GAAP ETR _{i,t-1}	1.105*** (0.373)							
	$\textit{Growth} \times \textit{GAAPETR}_{i,t\text{-}1}$	(0.0.0)	-0.292						
	$\textit{Mature} \times \textit{GAAPETR}_{i,t-1}$		(0.201)	-0.491					
	Decline \times GAAP ETR _{i,t-1}			(0.302)	1.680***				
	Observations Stage-specific life cycle dummy Model controls Year fixed effects Industry fixed effects Std. error clustered by	90,790 YES YES YES FIRM	90,790 YES YES YES FIRM	90,790 YES YES YES YES FIRM	(0.382) 90,790 YES YES YES YES FIRM				
	PANEL B: Results after the inclusion o	f 3-year long run þ	performance using	GAAP ETRVol (Logit Model)				
	Omitted variable – Past negative perfor 1.laggedEBIT_TA	mance -3.335*** (0.179)	-3.652^{***} (0.176)	-3.494^{***} (0.175)	-3.473*** (0.175)				
	2.laggedEBIT_TA 3.lagged EBIT_TA	-0.553^{***} (0.158) -1.211^{***} (0.159)	-0.574^{***} (0.157) -1.311^{***} (0.158)	-0.538^{***} (0.157) -1.211^{***} (0.159)	-0.552^{***} (0.157) -1.284^{***} (0.159)				
	Life cycle \times Tax avoidance as GAAP E GAAP ETRVol _{i,(t-1,t-5)}	CTR volatility 2.994*** (0.423)	2.850*** (0.432)	2.322*** (0.430)	3.110*** (0.408)				
	Life cycle \times Tax avoidance Introduction \times GAAP ETRVol _{i,(t-1,t-5)}	-1.918^{***}							
	$\textit{Growth} \times \textit{GAAP ETRVol}_{i,(t\text{-}1,t\text{-}5)}$	(0.700)	0.190						
	$Mature \times GAAP ETRVol_{i,(t-1,t-5)}$		(0.550)	1.322**					
Table S6.	$Decline imes GAAP ETRVol_{i,(t-1,t-5)}$			(0.547)	-3.833***				
Addressing endogeneity from omitted variable bias: past negative	Observations Stage-specific life cycle dummy	90,790 YES	90,790 YES	90,790 YES	(1.104) 90,790 YES				
performance					(continued)				

		Der	fault		Tax planning
	(1)	(2)	(3)	(4)	and financial
Model controls	YES	YES	YES	YES	default
Year fixed effects	YES	YES	YES	YES	
Industry fixed effects	YES	YES	YES	YES	
Std. error clustered by	FIRM	FIRM	FIRM	FIRM	
Note(s): Table S6 Panel A shows the <i>GAAPETR</i> as independent variable at the results obtained by running the variable and including a control for p introduction stage and including a measure are tage and including a measure the stage and incl	e results obtained and including a con life cycle-wise log ast performances. I easure of 3-year lag of 3 year lagged p	by running the lift trol for past perfor it regression usin in both the cases, (ged performance.	e cycle-wise logit : rmances. Table S0 g <i>GAAP ETRVol</i> Column 1 reports Column 2 reports	regression using 6 Panel B shows 7 as independent the results at the the results at the sults obtained at	355

the mature stage and including a measure of 3-year lagged performance. Column 4 reports the results at the decline stage and including a measure of 3-year lagged performance. 3-year lagged firm performance is proxied by using lagged values of *EBIT_TA* other than the 1-year lagged *EBIT_TA* already included in the model as a control variable. All the variables are winsorized at 1 and 99% to avoid the impact of outliers. Robust standard errors in parentheses. ***p < 0.01, **p < 0.05, *p < 0.1Source(s): Created by authors

Table S6.

Corresponding author

Alessandro Gabrielli can be contacted at: alessandro.gabrielli@unipi.it

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com